# The Bureaucratic Challenge to AI Governance: An Empirical Assessment of Implementation at U.S. Federal Agencies

Christie Lawrence Stanford Law School Harvard Kennedy School Stanford, California, USA christie.lawrence@stanford.edu Isaac Cui Stanford Law School Stanford, California, USA iqcui@stanford.edu Daniel E. Ho Stanford University Stanford, California, USA dho@law.stanford.edu

### ABSTRACT

Can government govern artificial intelligence (AI)? One of the central questions of AI governance surrounds state capacity, namely whether government has the ability to accomplish its policy goals. We study this question by assessing how well the U.S. federal government has implemented three binding laws around AI governance: two executive orders-concerning trustworthy AI in the public sector (E.O. 13,960) and AI leadership (E.O. 13,859)-and the AI in Government Act. We conduct the first systematic empirical assessment of the implementation status of these three laws, which have each been described as central to US AI innovation. First, we track, through extensive research, line-level adoption of each mandated action. Based on publicly available information, we find that fewer than 40 percent of 45 legal requirements could be verified as having been implemented. Second, we research the specific implementation of transparency requirements at up to 220 federal agencies. We find that nearly half of agencies failed to publicly issue AI use case inventories-even when these agencies have demonstrable use cases of machine learning. Even when agencies have complied with these requirements, efforts are inconsistent. Our work highlights the weakness of U.S. state capacity to carry out AI governance mandates and we discuss implications for how to address bureaucratic capacity challenges.

### **CCS CONCEPTS**

• Social and professional topics  $\rightarrow$  Government technology policy.

### **KEYWORDS**

AI policy, policy implementation, regulation, bureaucracy

### **ACM Reference Format:**

Christie Lawrence, Isaac Cui, and Daniel E. Ho. 2023. The Bureaucratic Challenge to AI Governance: An Empirical Assessment of Implementation at U.S. Federal Agencies. In AAAI/ACM Conference on AI, Ethics, and Society (AIES '23), August 8–10, 2023, Montréal, QC, Canada. ACM, New York, NY, USA, 51 pages. https://doi.org/10.1145/3600211.3604701

AIES '23, August 8-10, 2023, Montréal, QC, Canada

© 2023 Copyright held by the owner/author(s).

ACM ISBN 979-8-4007-0231-0/23/08.

https://doi.org/10.1145/3600211.3604701

### **1** INTRODUCTION

Can government govern AI? Many commentators have discussed the *normative* question of government intervention into the market [80, 86, 93, 98, 103]. We address a distinct, but related, empirical question that highlights the bureaucratic challenge to AI governance: Is there sufficient *state capacity* to achieve the goals of AI governance when such goals have already been set in law?

Many scholars, policymakers, and commentators point to the transformative potential of AI [23, 125]. Seeking to capture the benefits of the "Fourth Industrial Revolution" or "third wave of the digital revolution," countries are prioritizing efforts to reorganize their public and private sectors, fund research and development (R&D), and establish structures and policies that unleash AI innovation [37, 73, 112, 143]. In the United States, the White House and Congress have promoted AI innovation and its trustworthy deployment by increasing R&D investments, exploring mechanisms to increase equitable access to AI-related resources through a National Artificial Intelligence Research Resource, funding National AI Research Institutes throughout the country, dedicating \$280 billion-through the CHIPS and Science Act-into domestic semiconductor manufacturing and "industries of tomorrow," and coordinating AI policy in the National AI Initiative Office within the White House [10, 32, 40, 49, 53, 54, 69, 72]. While many have rightly applauded the Blueprint for an AI Bill of Rights and the associated actions across the federal government [44, 48], implementing that framework ultimately requires government agencies convert guidance and principles into practice.1

Federal AI initiatives raise at least three interrelated questions that are relevant to the academic literature and that implicate questions about policy effectiveness. First, as a question of *regulatory paradigm*, we might ask about the proper role of the state vis-à-vis industry and civil society actors, especially given the deep information asymmetries that plague state-based regulatory initiatives [98]. Second, conditional on the chosen paradigm, we might also ask about the proper *policy instrument*, implicating familiar debates over the specificity of rules in comparison to standards [82, 96] or the proper target of rules [90] given the policy context. Third, after policy instruments are determined, we could assess the *capacity* of bureaucracies to effectuate those actions' purpose [81, 115].

We contribute to this scholarship through a systematic assessment of the federal government's progress in implementing three important binding laws that are seen as central to U.S. leadership in trustworthy AI.<sup>2</sup> Through extensive research, we study (i) the AI in

Permission to make digital or hard copies of part or all of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for third-party components of this work must be honored. For all other uses, contact the owner/author(s).

<sup>&</sup>lt;sup>1</sup>Similarly, the NIST AI Risk Management Framework, released in January 2023, is helpful guidance but must be voluntarily adopted [55].

<sup>&</sup>lt;sup>2</sup>For discussion of U.S. federal AI policy documents from 2016-20, see [129]

Government Act of 2020 [19, 28], which aimed to provide resources and guidance to federal agencies on AI; (ii) the Executive Order on AI Leadership (E.O. 13,859) [12], which mandated government-wide efforts to promote AI R&D, AI competitiveness, and public trust; and (iii) the Executive Order on Trustworthy AI in Government (E.O. 13,960) [25], which encouraged government adoption of AI to benefit the public and promulgated trustworthy AI principles.<sup>3</sup> Collectively, the AI in Government Act, the AI Leadership Order, and the Trustworthy AI Order are critical pillars to the U.S. strategy on AI<sup>4</sup> and to envisioning an ecosystem where the U.S. government leads in AI and promotes trustworthy AI [71].

While much progress has been made, our findings-from a systematic examination conducted between late October and mid-November 2022-are sobering and highlight longstanding concerns about bureaucratic capacity. The goal of these laws to foster a responsible AI innovation ecosystem is threatened by weak and inconsistent implementation across the administrative state. First, fewer than 40 percent of all 45 requirements across the three pillars could be publicly verified as implemented at the time of our examination, including major requirements to advance AI innovation and trustworthy AI. Second, the implementation of Agency AI Plans, which are intended to provide information about the agency's approach to AI regulatory activities and to foster the agency's strategic planning around AI, has been poor. Around 88 percent of agencies that are likely subject to the requirement to submit Agency AI Plans under the AI Leadership Order failed to do so by late 2022.<sup>5</sup> Third, roughly half or more of agencies had not published an inventory of AI use cases, as required under the Trustworthy AI Order and in contradiction with public transparency efforts. Given Congress has since made disclosing AI use case inventories a statutory requirement under the 2023 National Defense Authorization Act [77], the lack of implementation is especially concerning.

These findings suggest a lack of bureaucratic capacity compounded by issues of policy ambiguity: Agencies lack the expertise, committed leadership, and sheer personnel to strategically plan for and prioritize AI, and compliance is hindered by vague mandates and reporting lines. We thus suggest three policy recommendations. First, centralized mandates must delineate (1) which agencies and sub-agencies must comply, (2) what "AI" applications are covered, and (3) how to interpret non-responses. This places agencies on notice about their obligations and facilitates public accountability. Second, if bureaucratic capacity is to blame, Congress must provide more resources for agencies to obtain adequate technical expertise. Third, senior leadership at the White House and at agencies is needed and senior personnel at agencies should treat these requirements not as boxes to tick but as opportunities for strategic planning around AI.

Our paper proceeds as follows. Section 2 discusses related scholarship in public administration, bureaucratic politics, and transparency initiatives for public sector AI. Section 3 provides background on the three binding laws we assessed. Section 4 discusses our methodology for systematically assessing the implementation of these laws. Section 5 provides detailed findings on the implementation of the AI Leadership Order, Trustworthy AI Order, and AI in Government Act. Section 6 examines the AI Leadership Order's requirement that agencies publish Agency AI Plans in detail across 41 agencies. Section 7 assesses the Trustworthy AI Order's requirement that agencies publish AI use case inventories across 220 agencies and narrower subsets of agencies. Section 8 discusses implications and limitations and Section 9 concludes.

### 2 RELATED WORKS

Our study of bureaucratic implementation of AI governance speaks to four bodies of research. First, our work relates to longstanding scholarship on state and bureaucratic capacity to achieve policy goals [84, 109, 115, 121]. Prior research shows that agency performance and the realization of White House-level political goals are frustrated by organizational capacity constraints, including insufficient leadership, staff, and resources. For example, Bolton, Potter, and Thrower [81] analyzed 22,000 regulations reviewed by the Office of Information and Regulatory Affairs (OIRA) within OMB and found that organizational capacity constraints, including vacant leadership positions, insufficient staff resources, and high workloads, hindered the president's ability to advance priority rules and inhibited OIRA's ability to carry out its mission.

In the AI space, agencies' struggle to attract and retain technical talent is a hurdle to the executive branch's ability to responsibly adopt and govern AI [88, 101, 102]. One estimate is that while 60% of new machine learning PhD graduates went into industry and 24% into academia, less than 2% went into government in 2020 [144]. Embedded AI expertise, as Engstrom, Ho, Sharkey, and Cuéllar [102] detailed and as other scholars noted (e.g., [88, 101, 128]), is critical for agencies' efficacy in designing, developing, and using AI tools to achieve their mission and subjecting AI tools to meaningful accountability. These concerns about bureaucratic capacity, in turn, can inform broader normative assessments of the federal government's current ability to promote trustworthy AI.<sup>6</sup>

Second, our research speaks to the central debate on the role of government in AI policy, where jurisdictions have diverged between taking a more "passive" role that gives space for industry self-regulation versus an "active" role through direct regulation (e.g., [98]). These debates imagine diverse roles for the state, whether as an interlocutor with industry to help develop best practice, a research funder, an adopter of responsible and trustworthy AI technologies, a direct regulator, or some combination of the above

<sup>&</sup>lt;sup>3</sup>We do not focus on the National AI Initiative Act of 2020, as the National AI Advisory Commission is statutorily tasked with tracking the Initiative's progress, nor on the Artificial Intelligence Training for the Acquisition Workforce Act, as its passage in October 2022 precludes meaningful assessment of its implementation. For more on the National AI Advisory Council, tasked with "advising the President and the National AI Initiative Office on topics related to the National AI Initiative," the creation of which was called by the National AI Initiative Act of 2020, see [56, 95].

<sup>&</sup>lt;sup>4</sup>The U.S. government does currently not have a "National AI Strategy" per se, but instead has a number of documents, including the three assessed in this Paper, that collectively provide strategic guidance. The National AI Initiative Office maintains a list of related legislation, executive orders, and strategy documents. See [71].

<sup>&</sup>lt;sup>5</sup>The requirement is in Section 6(c) of the AI Leadership Order, [12], and OMB's guidance was published in a memorandum known as "OMB M-21-06" [138].

<sup>&</sup>lt;sup>6</sup>Cf., for example, Oxford Insight's AI Government Readiness Framework [133], which assesses government AI readiness in terms of, among other indicators, "[d]igital and data skills within government."

[86, 93, 103, 136]. Such normative debates can and should be informed by empirical evidence, including about the relative advantages and capabilities of different institutional actors. For example, Black and Murray [80] comment that a central issue about who ought to regulate concerns where "trust and legitimacy" lie whether for a transnational standard-setting organization, a corporation engaging in self-regulation, or a state-based regulatory body. Regulation has classically been justified based on the expertise of technocratic government agencies (e.g., [123]), but AI poses extreme information asymmetries between technology developers and policymakers [88, 91, 108], in addition to concerns about public-private gaps in expertise [88, 144]. For those who believe in a robust role for the state in AI governance, our work addresses a core question: whether the government has the capacity to effectively regulate AI.

Third, our work pertains to efforts for transparency around the administrative state [89, 101, 111]. Principles of transparency and accountability are foundational to administrative law (e.g., [87, 101]). In the U.S. context, much scholarship has examined transparency initiatives such as the Freedom of Information Act, sunshine laws and hearing requirements, notice-and-comment rulemaking, and public availability of agency guidance (e.g., [87, 107, 119, 127]). Calls for greater transparency around the U.S. government's use of AI are therefore situated not only within research about the role of transparency in administrative law but also within discussions about the benefits and risks posed by agencies' use of AI (see, e.g., [88, 102]). One major question surrounds how public sector AI challenges administrative law's commitment to transparency. Coglianese and Lehr [92] argue that the opacity of AI does not pose particular barriers to administrative law. Engstrom and Ho [100], on the other hand, argue that existing administrative law doctrines may be insufficient, requiring adaptations of governance. The importance of government transparency about its use of AI necessitates a discussion about the proper lever to achieve such transparency.7

Last, many efforts have focused on transparency through public registries of AI use cases. Floridi [104] discusses the promise of AI registries in Helsinki and Amsterdam, noting that, the "goal is to make the use of urban AI solutions as responsible, transparent, and secure as other local government activities." Other countries, such as the United Kingdom, have adopted these AI registries [116]. At the local government level in the U.S., Bloomberg Philanthropies uses AI registries as one evaluation criterion for its "What Works Cities" Certification, which it claims is the "national standard of excellence for data-driven, well-managed local government" [68]. The City of San Jose, for instance, began an Algorithm Register in January 2023 for transparency of city services [60]. Yet the implementation of such transparency initiatives has not been straightforward. New York City's Automated Decision Systems Task Force fractured in substantial part because of a lack of consensus around what constituted algorithmic decision systems. Cath and Jansen [85] question the efficacy of the Helsinki and Amsterdam model of AI registries as a form of governance. The Administrative Conference of the United States (ACUS) commissioned a report that compiled AI use

cases across federal regulatory agencies [102], requiring a large team to determine, for instance, whether the underlying use case met a definition of machine learning. This report preceded the promulgation of the AI Use Case Inventory requirement via executive order. And because requirements differ across jurisdictions, efforts like the Northwestern Computational Journalism Lab's Algorithm Tips have attempted to crowdsource information across the federal, state, and local level [62]. AI registers have been advocated in other domains as well [132], and remain one of the critical levers for transparency. Our research examines the actual implementation of such AI registries and demonstrates that substantial policy guidance may be required for faithful implementation.

### **3 LEGAL SETTING**

We address this core question of bureaucratic capacity for AI governance by assessing three pillars of America's strategy for AI innovation. The two executive orders and AI in Government Act all carry the force of law, and so the executive branch's ability to implement them serves as an important litmus test for the U.S. government's realization of its AI policy goals. Moreover, these laws are billed as cornerstones of America's AI policy. By enabling America "to coordinate AI strategy" and equipping federal agencies' responsible use of AI, the AI in Government Act sought to ensure America's "competitive edge against the rest of the world in the next decade" [28]. The AI Leadership Order was similarly touted as "critically important to maintaining American leadership in technology and innovation" [16], whereas the Trustworthy AI Order "signal[ed] to the world" America's commitment to "the development and use of AI underpinned by democratic values" [11, 24]. To achieve their stated goals, the AI Leadership Order sought to drive technological breakthroughs throughout all sectors of the U.S., while the two other efforts focused on the federal government's use of AI. We describe each of the laws in turn.

Executive Order 13,859 (The AI Leadership Order). The 2019 AI Leadership Order launched the American AI Initiative to "focus the resources of the Federal government to develop AI in order to increase our Nation's prosperity, enhance our national and economic security, and improve quality of life for the American people" [11]. Specifically, it sought to accelerate the federal government's efforts to build the infrastructure, policy foundations, and talent necessary for America's leadership in AI through a multipronged approach emphasizing AI R&D, AI-related data and resources, regulatory guidance and technical standards, the AI workforce, public trust in AI, and international engagement [11, 12, 117]. Noting that a "coordinated Federal Government strategy" was necessary and that AI "will affect the missions of nearly all executive departments and agencies," the AI Leadership Order further mandated that agencies pursue six related strategic objectives for "promoting and protecting American advancements in AI." These six strategic objectives were about: (1) investing in AI-related research and development; (2) making AI resources (e.g., data, models, computing resources) available to the public; (3) reducing barriers that prevent the development and use of AI technologies; (4) ensuring that domestic and international technical standards "minimize vulnerability to attacks from malicious actors and reflect Federal priorities"; (5) building the AI workforce; and (6) developing a National Security Presidential Memorandum "to protect the advantage of the United States in

<sup>&</sup>lt;sup>7</sup>Calls for transparency exist not only at the federal level but also at the state level. A proposal in California (A.B. 331), for example, seeks to require AI developers to submit impact assessments annually to the California Civil Rights Department [142].

Christie Lawrence, Isaac Cui, and Daniel E. Ho

AI and technology critical to United States economic and national security interests" [12].

Executive Order 13,960 (The Trustworthy AI Order). The 2020 Trustworthy AI Order directed federal agencies to harness "the potential for AI to improve government operations" [24]. Recognizing that "[t]he ongoing adoption and acceptance of AI will depend significantly on public trust," the Trustworthy AI Order articulated nine principles for federal agencies to implement-according to guidance that would be developed by the OMB-when designing, developing, acquiring, and using AI. These principles provide that AI should be (a) lawful, (b) performance-driven, (c) accurate, reliable, and effective, (d) safe, secure, and resilient, (e) understandable, (f) responsible and traceable, (g) regularly monitored, (h) transparent, and (i) accountable [25]. To support federal AI adoption, it also mandated several actions intended to increase the number of federal employees with necessary AI implementation expertise [24]. Like the AI Leadership Order, the Trustworthy AI Order required agencies to publicly disclose certain AI-related information in an attempt to cultivate trust and understanding (see Section 7). The requirement of disclosing AI use cases was also incorporated into the 2023 National Defense Authorization Act [77], meaning Congress, too, has directed federal agencies to take inventory and disclose their uses of AI, reflecting the perceived importance of this transparency measure.

AI in Government Act of 2020. The AI in Government Act sought to "ensure that the use of AI across the federal government is effective, ethical and accountable by providing resources and guidance to federal agencies" [28]. This included the establishment of an AI occupational series, a call for formal guidance for agency usage, procurement, bias assessment and mitigation of AI, and the creation of a center of excellence within the General Services Administration (GSA) to support government adoption of AI.

### 4 METHODOLOGY

These three laws have been in effect for sufficient time to enable us to design a study to assess the implementation status of each line-level provision. The research was based on an extensive manual search protocol—conducted between October and November of 2022—detailed in Appendices A.1, B.1, and C.1, but we provide a concise overview of our research approach here. We note at the outset that because these laws impose public transparency and reporting requirements, we rely on public materials to conduct our searches. We undertook extensive efforts to identify relevant documents or notices of actions, but these may not capture all relevant (nonpublic) actions. Our findings still remain informative about the transparency of national AI efforts, and failures to implement by statutory or regulatory deadlines are particularly informative.

To assess overall implementation, we identified all line-level actions within the three documents (e.g., instructions that a federal entity "shall budget," "shall consider," "shall review," "shall publish"). Each line-level action was categorized as a *time-boxed requirement*, where the action was required by a specified date (e.g., publishing a report within 90 days); an *open-ended requirement*, where the mandated action did not have a specific date for completion; or an *ongoing requirement*, where the mandate did not include a specific deliverable or concrete outcome and where there was no specified deadline. It was generally straightforward to assess whether the time-boxed requirements were met, whereas other mandated actions were often more ambiguous, either due to lack of a deadline, lack of express public disclosure requirements, or both. We construed ambiguity in favor of the agencies based on an assumption that the agencies were taking the necessary steps (or at least making good-faith efforts) to implement these mandates, as explained in Appendix A.

In addition, we studied the implementation status of two specific cross-agency mandates: the requirement under the AI Leadership Order for agencies to issue "Agency AI Plans," and the requirement under the Trustworthy AI Order for agencies to post AI use case inventories. In the former, the AI Leadership Order required "implementing agencies"—defined to be agencies, as determined by the National Science and Technology Council (NSTC) Select Committee on AI, with regulatory authorities and that "conduct foundational AI R&D, develop and deploy applications of AI technologies, provide educational grants, and regulate and provide guidance for applications of AI technologies"—to issue a report discussing its authorities and plans to regulate AI. The Trustworthy AI Order, by contrast, ordered all "agencies" (with exceptions only for military, intelligence, and independent regulatory agencies) to disclose their uses of AI.

Ambiguities in the scope of these executive orders—the agencies they cover and, for the AI use case inventories, the definition of "AI"—complicated assessment of their implementation. For the Agency AI Plans, we looked to agencies with regulatory authority and therefore included cabinet-level departments and agencies and the 19 agencies deemed "independent regulatory agencies" under 44 U.S.C. § 3502(5). We included the U.S. Agency for International Development (USAID), as it was the only agency represented at the National Security Council that was not already included as a Cabinet-level agency or as an independent regulatory agency. We also inquired with a member of the Select Committee and did not receive an answer on what agencies are included. The result was 41 agencies.

Because the AI Use Case Inventories requirement applied to agencies generally, we began with the Administrative Conference of the United States' *Sourcebook of U.S. Executive Agencies* ("ACUS Sourcebook"). From the 278 agencies identified in the ACUS Sourcebook's data spreadsheet, we removed agencies within the Department of Defense, agencies and sub-agencies within the intelligence community, and the 19 independent regulatory agencies defined in 44 U.S.C. § 3502(5), based on the exemptions in Section 8(a) of the Trustworthy AI Order. This left us with a total of 220 agencies.

In searching for AI Plans and Use Case Inventories, we took a systematic approach meant to optimize both the chance of finding the document while also providing clear and simple search processes. For each requirement and agency, we searched in four ways: (1) at a dedicated URL as mandated under the respective executive order; (2) a web search for certain words closely related to the requirements; (3) a search on the agency's website for those key words; and (4) a search in the publication libraries at AI.gov.

Full data generated by our research included in the Appendices.

### **5 OVERALL IMPLEMENTATION STATUS**

While much progress has been made, we were unable to verify implementation of the majority of the line-level legal requirements.

	Implemented	Unknown	Not Implemented
AI Leadership Order	39%	57%	4%
Trustworthy AI Order	13%	75%	13%
AI in Government Act	17%	17%	67%
Total	27%	58%	16%

Table 1: Summary of Implementation as of Nov. 2022.

Across both executive orders and the AI in Government Act, we found that 11 of 45 requirements, or roughly 27 percent, were implemented (see Table 1).<sup>8</sup> The implemented requirements spanned a range of topics, including agencies' prioritization of AI R&D in annual budget proposals,<sup>9</sup> recommendations for leveraging cloud computing resources for federally funded AI R&D,<sup>10</sup> guidance on federal engagement in the development of AI-related technical standards,<sup>11</sup> and the establishment of a GSA AI Center of Excellence to facilitate the adoption of AI within the federal government.<sup>12</sup>

However, seven of 45 requirements (16 percent) were not implemented by the deadline, and the remaining 26 requirements (58 percent) could not be confirmed as either fully implemented or not implemented (see Appendix A.2). The requirements that remain unfulfilled—including creating an AI occupational series for federal employees, estimating the AI workforce gap in the federal government, policy guidance on federal acquisition and use of AI,<sup>13</sup> and a public roadmap on OMB's intended revisions or new AI policy guidance<sup>14</sup>—are significant for the country's AI ecosystem and the

<sup>10</sup>Section 5 of the AI Leadership Order [12] directs the Secretaries of Defense, Commerce, Health and Human Services, and Energy, as well as the Administrator of NASA and the Director of the NSF, to prioritize allocation of high-performance computing resources for AI, and also directs the NSTC Select Commission on AI to work with GSA on a report to the president for leveraging cloud computing resources. The National AI Initiative Office's AI Researchers Portal includes a computer resources overview with six "Federally-supported computing infrastructure resources that are useful for AI research" identified. See [59]. The NSTC Select Committee on AI also published—16 months after the mandated deadline—*Recommendations for Leveraging Cloud Computing Resources for Federally Funded Artificial Intelligence Research and Development* as well as a complementary "lessons learned" report in July 2022. See [26, 52].

<sup>11</sup>Section 6(d) of the AI Leadership Order [12] directs the Secretary of Commerce through the NIST Director, with participation from relevant agencies, to "issue a plan for Federal engagement in the development of technical standards and related tools in support of reliable, robust, and trustworthy systems that use AI technologies." In August 2019, NIST published the required report. See [17].

federal government's adoption of AI. Similarly, the implementation status is uncertain for major requirements, including efforts to make data and source code more accessible for AI R&D,<sup>15</sup> better leverage and create new AI-related education and workforce development programs,<sup>16</sup> and ensure agencies participate in interagency bodies that further the implementation of trustworthy AI.<sup>17</sup>

Requirements in the executive orders with deadlines for specific deliverables were implemented at a higher rate. Conversely, none of the AI in Government Act's four requirements with a deadline were implemented: the Office of Personnel Management (OPM) was to submit to Congress a plan to establish an AI occupational series by May 2021; OMB was required to issue a memorandum on AI procurement, mitigating discriminatory impact or bias, and promoting AI innovation by October 2021, with agencies publicly posting plans to achieve consistency with it by April 2022; and OPM was to create an AI occupational series and estimate AI-related workforce needs in each federal agency by July 2022. Of the implemented requirements across all three, many were late. For example, the NSTC Select Committee on AI produced the AI Leadership Order's mandated report to the president on better leveraging cloud computing for AI about 16 months past the deadline. Pursuant to the AI Leadership Order, OMB similarly issued a memorandum to agencies on regulatory approaches to AI about 16 months late, as well as a notice on the Federal Register soliciting public comments on how to improve public access to federal data for AI about two months after the AI Leadership Order's summer 2019 deadline.

We provide detailed findings in Appendix A.2 and a line-level tracker in Appendix E.1.

### 6 AGENCY AI PLANS

As noted above, a significant focus of the AI Leadership Order was "reduc[ing] barriers to the use of AI technologies to promote their innovative application" while also protecting "civil liberties, privacy, American values, and United States economic and national security" [12]. The AI Leadership Order therefore placed significant emphasis on examining the proper role of regulating AI, noting the desire to "avoid regulatory or non-regulatory actions that needlessly hamper AI innovation and growth" [138].

Two requirements were critical to achieving this objective: (1) OMB was required to publish a memorandum providing guidance on how agencies should approach regulating AI, and (2) agencies

<sup>&</sup>lt;sup>8</sup>A requirement in Section 5(c)(ii) of the Trustworthy AI Order [25] had not been implemented when we did our systematic analysis, but we excluded this requirement from the overall implementation assessment because the deadline for its implementation had not yet passed.

<sup>&</sup>lt;sup>9</sup>Section 4(a) of the AI Leadership Order [12] directed heads of AI R&D agencies to "consider AI as an agency R&D priority" and to take AI "into account when developing budget proposals and planning for the use of funds." Section 4(b) directed the same agencies to "budget an amount of AI R&D that is appropriate for this prioritization," particularly through the Networking and Information Technology Research and Development (NITRD) Program, and to identify "the programs to which the AI R&D priority will apply and estimate the total amount of such funds that will be spent on each program." This ongoing, annual requirement seems to be implemented through an annual NITRD supplement to the president's budget, progress reports on AI R&D, and a NITRD AI R&D dashboard. See [10, 13, 14, 22, 35, 64].

<sup>&</sup>lt;sup>12</sup>Section 103 of the AI in Government Act [19] mandates the establishment of this Center and delineates its roles; GSA has established the Center. See [139].

<sup>&</sup>lt;sup>13</sup>The White House announced in May 2023 that OMB will release draft guidance on AI procurement for federal agencies in summer 2023 [114].

<sup>&</sup>lt;sup>14</sup>These are respectively required by Sections 105 and 104 of the AI in Government Act [19] and Section 4(b) of the Trustworthy AI Order [25]. Note that Action 7 in the 2021 Federal Data Strategy Action Plan could arguably be construed as an implementation of the public roadmap requirement because it provides four milestones; however, it

does not mention policy guidance documents (e.g., OMB Circulars) as anticipated by the Trustworthy AI Order. See [34, p. 14].

<sup>&</sup>lt;sup>15</sup>Required under Section 5 of the AI Leadership Order [12].

<sup>&</sup>lt;sup>16</sup>Section 7 of the AI Leadership Order [12] mandates that the NSTC Select Committee on AI "shall provide recommendations to NSTC Committee on STEM Education regarding AI-related educational and workforce development considerations" an "provide technical expertise to the National Council for the American Worker." Furthermore it directs agencies to annually communicate plans to the NSTC Select Committee on AI about AI-related fellowship and service programs. Section 7 of the Trustworthy AI Order [25] mandates that OPM "shall create an inventory of Federal Government rotational programs and determine how these programs can be used to expand the number of employees with AI expertise" and "issue a report with recommendations" for doing so that is "shared with the interagency coordination bodies… enabling agencies to better use these programs for the use of AI…"

<sup>&</sup>lt;sup>17</sup>Section 6 of the Trustworthy AI Order [25] notes that agencies "are expected to participate in interagency bodies for the purpose of advancing the implementation of the Principles and the use of AI consistent with this order" and that the CIO Council "shall publish a list of recommended interagency bodies and forums in which agencies may elect to participate, as appropriate and consistent with their respective authorities and missions" to fulfill the expectation that they participate in interagency bodies to advance the AI principles.

with "regulatory authorities" were required to publicly post Agency AI Plans to "achieve consistency" with OMB's guidance. OMB's *Memorandum for the Heads of Executive Departments and Agencies on Guidance for Regulation of Artificial Intelligence Application* (OMB M-21-06 [138]), published on November 17, 2020 (about 16 months after the deadline), fulfilled the first requirement and urged a "regulatory approach that fosters innovation and growth and engenders trust, while protecting core American values." This "OMB AI Regulation Memo" described "policy considerations" to guide AI development. It (1) provided ten "principles for the stewardship of AI applications" to guide agencies,<sup>18</sup> (2) identified alternatives to regulation,<sup>19</sup> and (3) proposed actions, such as public communications and supporting voluntary consensus standards, that agencies could take to reduce barriers to the use of AL.<sup>20</sup>

The OMB AI Regulation Memo also provided guidance on Agency AI Plans. It required agencies to identify (a) their statutory authorities to regulate AI, (b) AI-related information that they were collecting on regulated entities, (c) statutory restrictions on their ability to collect or share such information, (d) regulatory barriers identified by stakeholder engagement, and (e) potential regulatory actions. Agencies were instructed to use an OMB-provided template, submit the plans by May 2021 (adhering to the AI Leadership Order's deadline), and publicly post their plans on their agency websites [138]. Critically, the memo did not provide guidance on which agencies were required to produce an Agency AI Plan: the AI Leadership Order's requirement applied to agencies with sufficient AI-related activities and "regulatory authorities," neither of which are self-defining or obvious.<sup>21</sup> We requested, but did not receive, information on the applicable agencies and have, as a result, approximated the relevant agencies as spelled out in the detailed methodology in Appendix B.1.

Out of 41 agencies assessed, only five (12 percent), posted an AI Plan using the template provided by the OMB AI Regulation Memo (see Table 2) by November 2022, even as the OMB AI Regulation Memo ordered agencies to publish them by May 2021. These agencies were the Departments of Energy (DOE), Health and Human Services (HHS), and Veteran Affairs (VA), as well as the Environmental Protection Agency (EPA) and USAID. Thirty-six agencies have

	Published	Not published	Total
Number of agencies	5	36	41
Percent	12%	88%	
			<b></b>

Table 2: Publication of Agency AI Plans as of Nov. 2022

not published an Agency AI Plan. The absence of plans published by the Departments of Transportation (DOT), Commerce (DOC), Homeland Security (DHS), and Housing and Urban Development (HUD) is notable, given what is commonly understood as within their regulatory and rulemaking purview and what sub-agencies fall under them. For instance, DOT includes the Federal Aviation Administration (FAA), governing civil aviation, and the National Highway Traffic Safety Administration, which administers federal motor vehicle safety standards.

Examination of the five Agency AI Plans also casts doubt on whether all agencies meaningfully attempted to identify relevant regulatory authorities. (We provide a detailed summary in Appendix B.2 of the substance of these five Agency AI Plans.) The DOE's AI Plan was completed with "None" written in every section. By contrast, HHS, the VA, and EPA provided more detail within their Agency AI Plan. Although the USAID plan does not identify any statutory authorities or planned regulatory actions, its publication of an Agency AI Plan demonstrates a commitment to transparency.

HHS is a particularly instructive and exemplary case. HHS identified 11 statutes that directly or indirectly authorized it to regulate AI applications, over 32 active collections of AI-related information, 12 AI use case priorities, 10 AI regulatory barriers, and four planned regulatory actions concerning AI applications [74]. The extent and depth of HHS's response likely stems from substantial efforts within the agency to formulate an AI strategic plan that considers how HHS will "[r]egulat[e] and oversee[] the use of AI in the health industry" as well as an extensive Trustworthy AI Playbook and an action plan by the Food and Drug Administration (FDA) for regulating AI-based medical devices [29, 31, 39]. In short, AI Plans reflect—and are aimed to foster—strategic planning, forethought, and coordination around AI.

### 7 AI USE CASE INVENTORIES

The Trustworthy AI Order mandated that agencies prepare inventories of their uses of AI, share them with the Federal Chief Information Officers Council (CIO Council) and other agencies, and then make them public [25]. The number of covered agencies is much broader than under the AI Leadership Order, exempting only independent regulatory agencies and agencies within the Department of Defense (DOD) or intelligence community.<sup>22</sup> Agency AI use case inventories must be prepared annually and should identify AI use cases that are inconsistent with the order, including the nine implementing principles. In the case of conflict, agencies are to develop remediation plans.<sup>23</sup> Guidance released by the CIO Council in fall 2021 explained that AI use case inventories were to be posted by March 2022 [75].

<sup>&</sup>lt;sup>18</sup>The principles were: (1) public trust in AI, (2) public participation, (3) scientific integrity and information quality, (4) risk assessment and management, (5) benefits and costs, (6) flexibility, (7) fairness and non-discrimination, (8) disclosure and transparency, (9) safety and security, and (10) interagency coordination. [138, pp. 3-7].

<sup>&</sup>lt;sup>19</sup>OMB M-21-06 provided four example non-regulatory approaches: (1) providing sector-specific policy guidance, statements, and frameworks; (2) using existing authorities to promote pilot programs and experimentation (e.g., through granting waivers, regulatory exemptions); (3) engaging voluntary consensus standards-development; and (4) developing and promoting voluntary frameworks. [138, pp. 7-8].

<sup>&</sup>lt;sup>20</sup>OMB M-21-06 suggested the following four "non-exhaustive" agency actions: (1) increase public "access to Federal data and models for AI R&D"; (2) public communication through requests for information (RFIs) in the *Federal Register*, increased transparency about uncertainties regarding outcomes, and making guidance documents widely available; (3) increase agency participation, including through private sector engagement, "in the development and use of voluntary consensus standards and conformity assessment activities" in order to "help agencies develop expertise in AI and identify practical standards for use in regulation"; and (4) increase international cooperation on regulation. See [138, pp. 8-11].

<sup>&</sup>lt;sup>21</sup>The Agency AI Plan requirement only applied to "implementing agencies" that have regulatory authorities, including independent regulatory agencies. "Implementing agencies" were defined in Section 3 of the AI Leadership Order [12] as "agencies that conduct foundational AI R&D, develop and deploy applications of AI technologies, provide educational grants, and regulate and provide guidance for applications of AI technologies, as determined by the co-chairs of the NSTC Select Committee."

 $<sup>^{22}</sup>$  The Trustworthy AI Order [25], in Section 8. For the specific language in the EO and our operationalization, see Section 4 and Appendix C.1.

<sup>&</sup>lt;sup>23</sup>As noted earlier, a similar AI use-case inventory requirement was adopted in the Fiscal Year 2023 National Defense Authorization Act [77].

Agencies	Org. Level	No inventories	Total	Perc.
All	Sub-agency	168	220	76%
	Parent	61	78	78%
Large	Sub-agency	78	125	62%
	Parent	21	37	57%
Known AI	Sub-agency	23	49	47%
	Parent	11	23	48%

Table 3: Publication of Agency AI Use Case Inventory as of Nov. 2022. "Large" agencies are those with more than 400 employees; "Known AI" are those with known AI use cases as of 2020. "Sub-agency" treats hierarchically related agencies as separate (e.g., separating the FAA and DOT); "Parent" attributes all sub-agency use cases to the parent agency.

Public disclosure of AI use case inventories has been problematic.<sup>24</sup> Roughly half or more of relevant agencies—a minimum of 47 percent of the agencies examined—have not published an AI use case inventory (see Table 3 and Appendices C.2 and E.3). Because of uncertainty in the relevant agencies, we report the implementation rate with different groups of agencies and at different organizational levels (see Appendix C.1 for more details on the methodology). The Trustworthy AI Order and the CIO Council's guidance for creating the inventories [25, 75], for instance, did not explain how sub-agencies and parent agencies should report their inventories (e.g., whether the DOT should include AI use cases from its subagency, the FAA, or let the FAA publish a separate inventory). We report use cases first with sub-agencies assessed individually and then rolled up to the parent agency.

Starting with the 220 agencies identified as potentially subject to this requirement—168 did not have an independent AI use case inventory or include their AI use cases within the inventory of their parent agency. Examining 78 parent-level agencies, only 17 posted AI use case inventories.<sup>25</sup> Thus, 76 percent of all 220 parent and sub-agencies, assessed separately, did not publish an inventory, and 78 percent of agencies assessed at the parent level did not publish an inventory (see Table 3).

To address the reality that executive agencies are not all similarly resourced, we also examined "large" agencies (defined as ones with over 400 employees). When focused on this subset of 125 large agencies (with parent and sub-agencies separately assessed), 47 had AI use cases published within an inventory, whereas 78 (62 percent) had not published use cases within an inventory. Assessing 37 large, parent-level agencies, 21 (57 percent) had not published an inventory.

The Trustworthy AI Order and guidance provided by the CIO Council did not specify whether an agency without AI use cases (or whose only use cases were exempted from disclosure) was required to file an inventory, or otherwise notify the public, to indicate that it had completed the requirement. It could be that 76 percent of agencies simply have no AI use cases. We hence examine the subset of agencies for which we can independently confirm the existence of AI use cases. This analysis enables us to distinguish whether the absence of inventories indicates the absence of AI use cases or an agency's failure to fulfill the Trustworthy AI Order's mandate. We rely on the extensive ACUS Report that "rigorous[ly] canvas[sed] AI use at the 142 most significant federal departments, agencies, and sub-agencies" to identify which agencies already had an AI use case as of 2019 and reported that nearly half of agencies have experimented with AI and machine learning at that time.<sup>26</sup> Of the 49 parent and sub-agencies with a known AI use case, 47 percent had not published an AI use case inventory (23 parent and subagencies). Among the narrowest group of agencies-i.e., 23 large agencies with a known AI use case assessed at the parent level-only 11 had published an AI inventory.<sup>27</sup> Notably HUD publicly disclosed that it does "not currently have any relevant AI use cases" [41]. We list these 23 agencies in Table 8. We also include an assessment of the implementation of the AI use case inventories of agencies enumerated in the Chief Financial Officers Act of 1990 and that are members of the CIO Council in Appendix C.2 and in Section 8.2.

The inventories themselves highlight serious implementation challenges with a signature transparency initiative. First, agencies are not disclosing AI use cases, even when these use cases have already been publicly documented. Customs and Border Protection (CBP), for instance, uses the Traveler Verification Service (TVS), which is a facial recognition system that "serves as CBP's backend matching service for the collection and processing of facial images in support of biometric entry and exit operations" [45, 102]. Acknowledging that "facial recognition poses a unique set of privacy issues" [9], CBP has sought to be "aggressively transparent" [45] in publishing privacy compliance documentation concerning its biometric entry-and-exit operations, including by publishing six Privacy Impact Assessments<sup>28</sup> and 13 Privacy Threshold Analyses, and building a public-facing website about the technology [45, 70]. While CBP has disclosed some uses of AI under the inventory posted by DHS, TVS is not among them [43].

<sup>&</sup>lt;sup>24</sup>We searched for AI use case inventories starting in late October 2022, and the findings reported in the Tracker are current up to at least November 11, 2022, with some spot checks performed throughout early December 2022. Agencies may have posted inventories after our exhaustive search. But they were required to post the inventories by March 2022. Moreover, though it is possible we missed some inventories, we emphasize that they ought to be easily accessible. The CIO's guidance "encouraged" agencies to publish their inventories on a specific URL [75], and the NAIIO's repository [57] ostensibly includes all of the published inventories. Even if agencies have published inventories elsewhere, there are shortcomings to their implementation of the order if they are not published according to these methods.

<sup>&</sup>lt;sup>25</sup>Three had zero use cases (HUD, NIST, and NSF), and a fourth (SSA) had only five use-cases. These are questionable, but for the purposes of the first two measures, we mark them as compliant solely from the posting of their inventories. In contrast, we count HUD as noncompliant when assessing against the identified AI use cases, i.e., the "Known AI Cases" of Table 3, because while its inventory asserts that the agency has no AI use cases, the ACUS Report identified a non-zero number of use cases. Neither NIST nor NSF were included in the "Known AI Cases" measure because the ACUS Report did not identify a use case from NIST, and NSF is not a "large" agency within the meaning of the report, and so neither is counted specially as compliant for one measure and noncompliant for another measure, unlike HUD. For further methodological discussion, see Appendix C.1.

<sup>&</sup>lt;sup>26</sup>[102, p. 6]. For two reasons, the "known AI use case" estimate is potentially very conservative. First, likely have expanded their use of AI — the ACUS Report was conducted three years before our assessment. Second, the ACUS Report defined "AI" as "machine learning, which train models to learn from data"—a narrower definition than that used in the Trustworthy AI Order. For methodological considerations, including extended discussion of the definitional differences, see Appendix C.1.

<sup>&</sup>lt;sup>27</sup>As HUD was identified as having a known AI use case in the ACUS AI Report, we do not include HUD's public disclosure of no AI use cases within the 11 agencies that have published an AI use case inventory. In contrast, we included HUD within the agencies that implemented the requirement in our measurement of all agencies and "large" agencies.

<sup>&</sup>lt;sup>28</sup>Such assessments are a legal requirement under Section 208(b) of the E-Government Act of 2002, see [134].

Second, inconsistencies in how agencies have implemented the AI use case inventories illustrate three sources of policy ambiguity.

(1) Non-response. For agencies that have not posted inventories, it is unclear whether they are asserting that they have no uses of AI or simply have not fulfilled the requirement. Of the published inventories, three-from HUD, the National Institute of Standards and Technology (NIST), and the National Science Foundation (NSF) [41, 50, 63]-state that their agencies have no AI use cases that meet the Trustworthy AI Order's requirements.

(2) Agency structure. All inventories except for NIST's were published at the parent-agency level (e.g., by DOC or DOE, rather than the NOAA or the Office of Electricity). But it is unclear whether unlisted sub-agencies within an inventory did not have relevant use cases or whether they were unresponsive to a presumed request for reporting by the parent agency. In some cases, the latter seems very likely.29

(3) AI definition. The definition of AI provided in the 2019 National Defense Authorization Act and incorporated into the Trustworthy AI Order is potentially quite broad, reaching among other things, any "artificial system" that "is designed to approximate a cognitive task" or that can "learn from experience and improve performance when exposed to datasets."30 The breadth of that definition may make compliance harder for agencies when classifying particular technologies as "AI" for the purposes of an inventory.<sup>31</sup> For example, NOAA identified 36 AI use cases, representing the vast majority of the DOC's 49 AI use cases. The rest of Commerce's AI inventory [46] includes zero uses from the parent agency, five from the International Trade Administration, two from the National Telecommunications and Information Administration (NTIA), one from the Minority Business Development Administration, and five from the U.S. Patent and Trademark Office, with NIST publishing a separate inventory [50]. Ambiguity may result from both the breadth of the definition of covered AI-which includes uses that are new and existing, standalone and embedded, procured and developed in-house by the agency-and the carve-outs for sensitive or classified uses of AI, AI used for national security purposes, AI "embedded within common commercial products," and AI R&D, as provided in Section 9 of the Trustworthy AI Order [25].

Third, AI use case inventories often incorporate existing transparency initiatives, but with significant variation. Agencies are best

positioned to know what records exist regarding each AI use case, and some have provided useful links to published documentation. For example, many use cases in the DHS inventory include links (e.g., to privacy impact assessments); some EPA, HHS, Department of the Interior (INT), DOC, and Department of Agriculture use cases include links to relevant publications; and some Department of Labor (DOL), INT, and Department of Justice (DOJ) use cases reference publicly available code.

#### DISCUSSION 8

We now discuss broader implications emerging from this study, as well as some limitations. First, empirically, our top-level finding is that implementation has been lacking, which we interpret through the lens of bureaucratic capacity and policy ambiguity. Second, methodologically, we discuss how social scientists can study policy implementation in a rigorous and systematic way based on our case studies.

### **Broader Implications** 8.1

Foundational theoretical work in bureaucratic capacity has argued that lower capacity can prevent effective implementation of hierarchically imposed policy obligations. Huber [115] attributed this possibility to inhibitions on the principal's ability to punish failures on the part of the agent when the agent lacks sufficient capacity to implement the directive. Other explanations focus on the multiplicity of tasks and principals that each agency has, which implies that the agency may shirk obligations that lack enforcement mechanisms [97]. Still others might argue that policy directives understood as far from the organization's core "turf" may seem peripheral or unimportant and are thus ignored [141]. All of these different explanations can shed light on the lackluster implementation of these AI directives: Agencies, by and large, lack the technical expertise and committed leadership necessary to effectively implement and prioritize regulatory principles promulgated by the White House or Congress.

Our findings also reveal substantial policy ambiguity that places more decision-making costs on agencies seeking to comply with the directives, thereby further hampering implementation. Central questions pertaining to the scope of transparency obligations-like the AI plans and inventories-were left ambiguous by the executive orders and White House-level guidance. Our findings emphasized two sources of ambiguity-ambiguity in defining "AI" and "agency."

On the former, the definition of "AI" used in the Trustworthy AI Order left substantial discretion to the agencies to categorize their use of technology. For example, the Order's exemptions for "AI embedded in common commercial products" and for "AI research and development activities" are ambiguous.<sup>32</sup> We found large inconsistencies in the kinds of use cases disclosed by agencies: compare, for example, NOAA's disclosure of 36 AI use cases pertaining to scientific research with CBP's non-disclosure of facial recognition systems used for biometric entry and exit operations (a system for which CBP has published an independent website, presumably because of the politically sensitive nature of the operation).

Additionally, both orders swept in broad terms, making it hard to know which entities were obligated to publish AI plans or use case

<sup>&</sup>lt;sup>29</sup>Consider DOE: In its inventory [58], DOE reports 45 use cases from three sub-offices: Brookhaven National Laboratory (one use case); the Office of Electricity (10 use cases); and Idaho National Laboratory (34 use cases). We think these numbers are implausible as an exhaustive account of AI usage within DOE. For example, a public information sheet published in 2020 from the then-Office of Fossil Energy (now the Office of Fossil Energy and Carbon Management) boasted of having "over 60 AI-enabled projects underway" [18]. Moreover, each DOE office has listed a single individual as its point of contact for all AI use cases from that office. It seems at least plausible that those offices have designated specific employees to serve as point-individuals on AI transparency for the office but that other offices have failed to do so, which is why there are no use cases reported for other DOE sub-agencies. As another example, in the Department of the Interior's inventory [47], the United States Geological Survey (a sub-agency) disclosed 55 of the Department's 65 use cases. Some of those use cases seem to be collaborations with agencies (e.g., the U.S. Fish and Wildlife Service and the Bureau of Ocean Energy Management) that themselves did not disclose use cases. We count such agencies as failing to implement the requirement notwithstanding that other agencies reported some of their AI use cases. <sup>30</sup>The full definition is provided in Section 238(g) of the FY2019 NDAA [8].

<sup>&</sup>lt;sup>31</sup>The CIO's 2021 FAQs and "Example AI Use Case Inventory Scenarios" guidance documents [75] provide some details beyond the statutory definition, but much of the work of classifying technologies as "AI" still falls on the agencies.

<sup>&</sup>lt;sup>32</sup>And they may not be normatively justifiable: an agency's reliance on ChatGPT may be quite important depending on how the technology is used, even if ChatGPT is a common commercial product.

inventories (see discussion in Section 4 and Appendices B.1 and C.1). Yet the problem of "agency" definition is not novel. As the authors of the ACUS Sourcebook note, "cataloging administrative agencies is difficult because so many varying definitions abound" ([130, p. 11]). The point is not that there is a correct definition—rather, it is that these pillars of America's AI strategy did not even attempt to address the issue, thereby shifting costs onto lower-level executive branch entities to determine whether they ought to comply.

From the perspective of change management, a key problem with such ambiguity is that it inhibits the policymaker from effectively communicating and directing change in conditions of fast-changing technology (e.g., [137], cf. [79, 96]). Tighter rule construction itself would be helpful so that agencies better understand when and how they must comply. But discretion will inevitably vest with line-level bureaucrats implementing policies on AI (cf. [122]). Though much theoretical literature has discussed bureaucratic resistance to hierarchically imposed requirements (e.g., [118]), evidence from the perspective of "street-level" bureaucrats argues that implementation failures are more often a result of insufficient capacity than ideological opposition [83], which accords with our findings.

The poor public availability of the Agency AI Plans, AI use case inventories, and other mandated items supports existing scholarship about the consistency by which agencies make public guidance documents [87]. Coglianese found that mandated agency guidance is inconsistently published, which keeps the public in the dark about important agency actions. Agencies, Coglianese argued, need internal management practices to ensure disclosure because legal requirements without incentives or consequences for non-disclosure will be insufficient to motivate agencies to disclose [87]. Transparency requirements strengthen government accountability efforts while also enabling federal agencies to have meaningful consultations with external stakeholders. But to actualize those policies will require more careful rule construction from the top, closer attention to bureaucratic capacity down the chain, and agency adoption of management strategies to systematically track, index, and publish guidance [99].

Our systematic assessment of agency implementation of these policies provides evidence for the inference that insufficient bureaucratic capacity has hampered the implementation of U.S. AI policy. We do not rule out possible alternative explanations. For example, agencies' incentives to faithfully implement policy directives may be tied to their assessment of those policies' durability [135], where policies promulgated by a president late in her term<sup>33</sup> may be perceived by the agencies as less imperative or even less legitimate. Similarly, agencies may have differential incentives to comply based on how central AI initiatives are to their core functions, especially as it implicates funding. Thus, for example, NOAA's substantial disclosure of AI use cases in its inventory might be understood as a kind of "bureaucratic entrepreneurship" [124], where the agency's work helped demonstrate to the public why it needed greater funding for AI-related initiatives (funding, incidentally, which it received [106]). But while there is more room for theoretical insight from studying variation within our findings, the top-level result is still

indicative of a general lack of bureaucratic capacity to implement AI policy.

Finally, our methodological contribution is to provide a transparent and systematic means for assessing policy implementation notwithstanding the conceptual ambiguities noted above. Our reliance on a mix of statutes, regulatory provisions, and materials by ACUS can inform subsequent efforts to assess policy that is addressed, generally, toward "agencies," as in the Trustworthy AI Order, or agencies with "regulatory authorities," as in the AI Leadership Order (see Section 4 and Appendices B.1 and C.1). Furthermore, for the AI use case inventories, we present findings with different levels of aggregation and groupings of agencies that correspond to theoretical concerns and practical realities: We considered not only the largest number of agencies to which the executive order might theoretically apply, but we also cut only to "large" agencies and to agencies that had been previously identified by ACUS as using (or considering using) AI. And we assessed each of those measures not only by disaggregating at the lowest "agency" level but also by bundling agencies into their parent departments (for example, including the IRS within the Treasury Department) to address what seems to be agencies' understandings of their obligation under the executive order (i.e., most disclosed inventories were housed at the parent-agency level) (see Section 7 and Appendix C). And while much of our work emphasized systematic analysis, we also considered the disclosures qualitatively so that our findings are sensitive not only to whether agencies ticked a box but also how meaningfully their disclosures achieved the executive orders' policy goals. While these steps involve nuance to implement, they illustrate how we can rigorously assess policy implementation. As AI governance efforts mature, these efforts will be critical to ensure that legislative and executive directives are not "lost or misdirected in the vast hallways of the federal bureaucracy" ([1, p. 1111]).

### 8.2 Limitations

We note several limitations of our assessment. First, as we have noted, our assessment is based on publicly available information. Many more implementation efforts may be underway. But the mere fact that so many deadlines have been missed—when the pace of innovation in AI is extremely fast—illustrates the severe limitations of existing governmental efforts. In addition, the difficulty in researching the implementation status is itself telling. Existing efforts have delegated to agencies the task of defining and implementing these provisions, and, as a result, efforts have been fragmented and inconsistent.

Second, some might argue that the failure to meet deadlines and implement legal requirements is no different in AI than in other domains [78]. Perhaps that is so, although there are few directly analogous studies in comparable, but non-AI, domains.<sup>34</sup> Regardless, our findings suggest bureaucratic capacity challenges in a highly consequential space.

Third, our implementation estimates may be critiqued based on the fact that they weigh provisions equally. Not all operative provisions in a bill or order matter equally. We agree and have provided the detailed, line-level tracker results to enable any assessment of

<sup>&</sup>lt;sup>33</sup>The AI Leadership Order was issued on February 14, 2019, and the Trustworthy AI Order on December 8, 2020, when then-President Trump was a so-called 'lame duck' president.

<sup>&</sup>lt;sup>34</sup>For an exception, see the Government Accountability Office's assessment of the implementation of the Open, Public, Electronic and Necessary Government Data Act of 2018 [36].

implementation of specific items (Appendix E). Our qualitative assessment, however, does not suggest that all important items have been implemented. To the contrary, major items that are critical to preparing the federal government for the AI transition have not been addressed.

Fourth, while AI use case inventories are an important step toward transparency, they remain relatively limited as implemented. Some registries, for instance, include extensive data and model documentation, but the Trustworthy AI Order did not appear to require such extensive detail. As we show in Appendix C.2, numerous agencies have gone beyond the minimal requirement and documented performance benchmarks and evaluation measures, which are particularly important for assessments of trustworthiness.

Last, we released our findings in December 2022 [120] and some agencies have since posted AI use case inventories or disclosed no use cases.<sup>35</sup> To the extent that our research galvanized agency action, we both applaud the agencies and White House for taking initiative but also re-emphasize that formal compliance itself is not the goal. Compliance should be a means for strategic planning and action: Publicly verifiable steps, while important from a transparency perspective, are fundamentally proxies for assessing whether agencies are prepared for and taking concrete steps around trustworthy AI. If the tracked metrics become ends in themselves, then they are no longer reliable indicators of the underlying issue of interest. Agency responses also suggest further support for our conclusion that senior leadership is critical. All of the agencies that we know published an inventory after our white paper, except for one, are subject to the Chief Financial Officers Act (see Appendix D). This act required each agency to establish a Chief Financial Officer and provided the White House's Office of Management and Budget (OMB) greater authority over agency financial management [67]. This could demonstrate the important role of the White House in shepherding compliance and strategic planning.

### 9 CONCLUSION

Our findings have broad implications for the current ability of government to govern AI. We find that three core elements of America's collective AI strategy—the AI Leadership Order, the Trustworthy AI Order, and the AI in Government Act—have not been implemented well despite an urgent need for the U.S. government to grapple with a technology that is widely seen to have far-reaching, transformative potential.

These findings strongly suggest that there is a resource shortage, a leadership vacuum, and a capacity gap, which are exacerbated by policy ambiguity. Leadership will be required from both the White House, including the National AI Initiative Office and OMB, and agencies to coordinate and drive forward AI innovation and trustworthy adoption. Current requirements may appear to agencies like "unfunded mandates" and be treated like checklists when they should in fact be seized as opportunities for strategic planning around AI. Some agencies have recognized the urgent need and were able to respond comprehensively and meaningfully to these legal requirements (see, e.g., HHS's Artificial Intelligence Strategy, Trustworthy AI Playbook, and action plan for regulating AI-based medical devices [29, 31, 39]). If our findings are due to bureaucratic capacity, Congress should provide resources for agencies to staff and acquire technical expertise to comply in more than a perfunctory way and develop strategic AI Plans. Failure to provide proper resources and mandate senior personnel to discharge these responsibilities could otherwise undermine the goal of these laws to maintain U.S. leadership in AI innovation and trustworthy AI.

The public disclosure of AI Plans and AI use case inventories constitute an important effort to foster transparency and accountability in public sector AI. The executive orders mandated their public disclosure and senior-level guidance instructed they be made readily available on specific websites. The fact that it has taken considerable effort for our team to track the implementation of such plans, use cases, and requirements (see efforts detailed in the Appendices) strongly suggests that improvements must be made on reporting and tracking of these provisions. Our assessment may miss certain use case inventories, for instance, but that is precisely the point. Disclosure must be accessible and legible to be effective.

We close by noting that on paper and in principle, America's strategy for AI innovation and responsible AI, as manifested in the Trustworthy AI Order, the AI Leadership Order, and the AI in Government Act, is highly laudable. But in practice, our assessment suggests severe challenges in the federal government's ability to navigate a rapidly changing and critically important space. Requirements have been converted into perfunctory checklists instead of triggers for strategic planning, and agencies do not appear to have effectively grappled with the opportunities and risks that AI poses.

Bureaucratic capacity is a sine qua non for turning laudable principles into reality.

### ACKNOWLEDGMENTS

We would like to thank Victoria Espinel, Janet Haven, Fred Oswald, Christine Tsang, Russell Wald, and Daniel Zhang for helpful comments. An earlier version of this study was released as a White Paper, "Implementation Challenges to Three Pillars of America's AI Strategy," from the Stanford Institute for Human-Centered AI (HAI) and Stanford RegLab.

### REFERENCES

- [1] 1971. Calvert Cliffs' Coord. Comm. v. A. E. Comn'n. , 1109 pages. https://casetext.com/case/calvert-cliffs-coord-com-v-a-e-comn
- [2] 2016. Objectives Report to Congress, Vol. 1, Fiscal Year 2017. Technical Report. National Taxpayer Advocate. https://www.taxpayeradvocate.irs.gov/wpcontent/uploads/2020/08/Volume\_1.pdf
- [3] 2017. Annual Assessment of the Internal Revenue Service Information Technology Program. Technical Report 2017-20-089. Treasury Inspector General for Tax Administration. https://www.tigta.gov/sites/default/files/reports/2022-02/201720089fr.pdf
- [4] 2017. The Return Review Program Increases Fraud Detection; However, Full Retirement of the Electronic Fraud Detection System Will Be Delayed. Technical Report 2017-20-080. Treasury Inspector General for Tax Administration. https://www.tigta.gov/sites/default/files/reports/2022-02/201720080fr.pdf
- [5] 2018. Annual Performance Report Fiscal Years 2017-2019. Technical Report. Social Security Administration. https://www.ssa.gov/budget/FY19Files/2019APR.pdf
- [6] 2018. Anti-Fraud Enterprise Solution. https://www.ssa.gov/privacy/pia/AFES\_ OPD%20Draft\_PIA\_04-17-2018.htm
- [7] 2018. Evaluation of the EEOC's Data Analytics Activites Final Report. Technical Report 2017-02-EOIG. Elder Research. https://oig.eeoc.gov/sites/default/files/ audits/EEOC%20Data%20Analytics%20Report%20Final%20.pdf
- [8] 2018. John S. McCain National Defense Authorization Act for Fiscal Year 2019.

<sup>&</sup>lt;sup>35</sup>Agencies that have since published an explanation or use case inventory include the Departments of Education, Housing and Urban Development, the Interior, the Treasury, and Transportation; the General Services Administration; the Small Business Administration; and the U.S. Office of Personnel Management.

- [9] 2018. Privacy Impact Assessment for the Traveler Verification Service. Technical Report DHS/CBP/PIA-056. Department of Homeland Security. https://www.dhs.gov/sites/default/files/publications/privacy-pia-cbp056tvs-january2020\_0.pdf
- [10] 2019. 2016-2019 Progress Report: Advancing Artificial Intelligence R&D. Technical Report. Artificial Intelligence Research & Development Interagency Working Group, Subcommittee on Networking & Information Technology Research & Development, Subcommittee on Machine Learning & Artificial Intelligence, and the Select Committee on Artificial Intelligence of the National Science & Technology Council. https://www.nitrd.gov/pubs/AI-Research-and-Development-Progress-Report-2016-2019.pdf
- [11] 2019. Accelerating America's Leadership in Artificial Intelligence. https://trumpwhitehouse.archives.gov/articles/accelerating-americasleadership-in-artificial-intelligence/
- [12] 2019. Maintaining American Leadership in Artificial Intelligence, Exec. Order No. 13,859, 84 Fed. Reg. 3967. https://www.federalregister.gov/documents/2019/ 02/14/2019-02544/maintaining-american-leadership-in-artificial-intelligence
- [13] 2019. The National Artificial Intelligence Research and Development Strategic Plan: 2019 Update. Technical Report. Select Committee on Artificial Intelligence of the National Science & Technology Council. https://www.nitrd.gov/pubs/National-AI-RD-Strategy-2019.pdf
- [14] 2019. The Networking & Information Technology Research & Development Program Supplement to the President's FY2020 Budget. Technical Report. Subcommittee on Networking & Information Technology Research & Development Committee on Science & Technology Enterprise of the National Science & Technology Council. https://www.nitrd.gov/pubs/FY2020-NITRD-Supplement.pdf
- [15] 2019. PIA ID Number 4592, Return Review Program. https://www.irs.gov/pub/ irs-pia/rrp-pia.pdf
- [16] 2019. Statement on executive order to maintain American leadership in artificial intelligence. https://beta.nsf.gov/news/statement-executive-order-maintainamerican
- [17] 2019. U.S. Leadership in AI: A Plan for Federal Engagement in Developing Technical Standards and Related Tools. Technical Report. National Institute of Standards and Technology. https://www.nist.gov/system/files/documents/2019/08/10/ai\_ standards\_fedengagement\_plan\_9aug2019.pdf
- [18] 2020. 5 Uses for Artificial Intelligence You've Never Heard Of. https://www.energy.gov/sites/prod/files/2020/04/f73/5%20Uses%20for% 20Artificial%20Intelligence.pdf
- [19] 2020. AI in Government Act of 2020, in the Consolidated Appropriations Act, 2021. https://www.congress.gov/116/plaws/publ260/PLAW-116publ260.pdf
- [20] 2020. Federal Chief Information Officers Council Charter. https://www.cio. gov/assets/files/CIOC-Charter-Dec-2020.pdf
- [21] 2020. National Taxpayer Advocate Annual Report to Congress 2020. Technical Report. National Taxpayer Advocate. https://www.taxpayeradvocate.irs.gov/ wp-content/uploads/2021/01/ARC20\_FullReport.pdf
- [22] 2020. The Networking & Information Technology R&D Program and the National Artificial Intelligence Initiative Office Supplement to the President's FY2021 Budget. Technical Report. Subcommittee on Networking and Information Technology Research and Development Committee on Science & Technology Enterprise of the National Science and Technology Council. https://www.nitrd.gov/pubs/ FY2021-NITRD-Supplement.pdf
- [23] 2020. OECD to host Secretariat of new Global Partnership on Artificial Intelligence - OECD. https://www.oecd.org/newsroom/oecd-to-host-secretariat-ofnew-global-partnership-on-artificial-intelligence.htm
- [24] 2020. Promoting the Use of Trustworthy Artificial Intelligence in Government. https://trumpwhitehouse.archives.gov/articles/promoting-usetrustworthy-artificial-intelligence-government/
- [25] 2020. Promoting the Use of Trustworthy Artificial Intelligence in the Federal Government, Exec. Order No. 13960, 85 Fed. Reg. 78939. https: //www.federalregister.gov/documents/2020/12/08/2020-27065/promoting-theuse-of-trustworthy-artificial-intelligence-in-the-federal-government
- [26] 2020. Recommendations for Leveraging Cloud Computing Resources for Federally Funded Artificial Intelligence Research and Development. Technical Report. Select Committee on Artificial Intelligence of the National Science & Technology Council, Washington, D.C. https://www.nitrd.gov/pubs/Recommendations-Cloud-AI-RD-Nov2020.pdf
- [27] 2020. Request for Comments on a Draft Memorandum to the Heads of Executive Departments and Agencies: Guidance for Regulation of Artificial Intelligence Applications, 85 Fed. Reg. 1825. https://www.federalregister.gov/documents/ 2020/01/13/2020-00261/request-for-comments-on-a-draft-memorandum-tothe-heads-of-executive-departments-and-agencies
- [28] 2020. Schatz, Gardner, Portman, Peters, Klobuchar Bill To Improve Federal Government's Use Of Artificial Intelligence Set To Become Law | U.S. Senator Brian Schatz of Hawaii. https://www.schatz.senate.gov/news/press-releases/schatzgardner-portman-peters-klobuchar-bill-to-improve-federal-governmentsuse-of-artificial-intelligence-set-to-become-law
- [29] 2021. Artificial Intelligence (AI) Strategy. Technical Report. Department of Health and Human Services. https://www.hhs.gov/sites/default/files/final-hhs-

ai-strategy.pdf

- [30] 2021. Artificial Intelligence (AI) Strategy. Technical Report. U.S. Department of Veterans Affairs. https://www.research.va.gov/naii/VA\_AI%20Strategy\_V2-508.pdf
- [31] 2021. Artificial Intelligence/Machine Learning (AI/ML)-Based Software as a Medical Device (SaMD) Action Plan. Technical Report. Food & Drug Administration Center for Devices & Radiological Health. https://www.fda.gov/media/145022/ download
- [32] 2021. The Biden Administration Launches the National Artificial Intelligence Research Resource Task Force | OSTP. https://www.whitehouse.gov/ostp/newsupdates/2021/06/10/the-biden-administration-launches-the-nationalartificial-intelligence-research-resource-task-force/
- [33] 2021. Department of State Enterprise Data Strategy: Empowering Data Informed Diplomacy. Technical Report. Department of State. https://www.state.gov/wpcontent/uploads/2021/09/Reference-EDS-Accessible.pdf
- [34] 2021. Federal Data Strategy: 2021 Action Plan. Technical Report.
- [35] 2021. The Networking & Information Technology R&D Program and the National Artificial Intelligence Initiative Office Supplement to the President's FY2022 Budget. Technical Report. Subcommittee on Networking and Information Technology Research and Develompment Committee on Science and Technology Enterprise and the Machine Learning and Artificial Intelligence Subcommittee Committee on technology Select Committee on Artificial Intelligence of the National Science and Technology Council. https://www.nitrd.gov/pubs/FY2022-NITRD-NAIIO-Supplement.pdf
- [36] 2021. Open Data: Additional Action Required for Full Public Access. Technical Report GAO-22-104574. United States Government Accountability Office. https: //www.gao.gov/assets/gao-22-104574.pdf
- [37] 2021. Remarks by National Security Advisor Jake Sullivan at the National Security Commission on Artificial Intelligence Global Emerging Technology Summit |NSC. https://www.whitehouse.gov/nsc/briefing-room/2021/07/13/remarksby-national-security-advisor-jake-sullivan-at-the-national-securitycommission-on-artificial-intelligence-global-emerging-technology-summit/
- [38] 2021. S&T Artificial Intelligence & Machine Learning Strategic Plan. Technical Report. Department of Homeland Security Science & Technology. https://www.dhs.gov/sites/default/files/publications/21\_0730\_st\_ai\_ml\_ strategic\_plan\_2021.pdf
- [39] 2021. Trustworthy AI (TAI) Playbook. Technical Report. Department of Health and Human Services. https://www.hhs.gov/sites/default/files/hhs-trustworthyai-playbook.pdf
- [40] 2021. The White House Launches the National Artificial Intelligence Initiative Office – The White House. https://trumpwhitehouse.archives.gov/briefingsstatements/white-house-launches-national-artificial-intelligence-initiativeoffice/
- [41] 2022. AI Inventory. https://www.hud.gov/data/AI\_Inventory
- [42] 2022. Annual Performance Report, Fiscal Years 2021-2023. Technical Report. Social Security Administration. https://www.ssa.gov/agency/performance/materials/ 2023/SSA\_FYs21-23\_APR\_Signed.pdf
- [43] 2022. Artificial Intelligence Use Case Inventory. https://www.dhs.gov/data/ AI\_inventory
- [44] 2022. Blueprint for an AI Bill of Rights: Making Automated Systems Work for the American People. Technical Report. The White House. https://www.whitehouse. gov/wp-content/uploads/2022/10/Blueprint-for-an-AI-Bill-of-Rights.pdf
- [45] 2022. CBP Privacy Evaluation (CPE) of the Traveler Verification Service (TVS) in support of the CBP Biometric Entry-Exit Program. Technical Report. U.S. Customs and Border Protection Privacy and Diversity Office. https: //www.cbp.gov/sites/default/files/assets/documents/2022-Sep/CPE%20Final% 20Report%20Traveler%20Verification%20Service%2020220815%20Final\_.pdf
- [46] 2022. DOC AI Use Case Inventory. https://www.commerce.gov/sites/default/ files/2022-09/DOC\_AI\_Use\_Case\_Inventory.pdf
- [47] 2022. DOI AI Use Case Inventory. https://www.doi.gov/sites/doi.gov/files/2021agency-inventory-of-ai-user-case-doi-20220318-submission-to-ombmax 0.xlsx
- [48] 2022. FACT SHEET: Biden-Harris Administration Announces Key Actions to Advance Tech Accountability and Protect the Rights of the American Public. https://www.whitehouse.gov/ostp/news-updates/2022/10/04/factsheet-biden-harris-administration-announces-key-actions-to-advance-techaccountability-and-protect-the-rights-of-the-american-public/
- [49] 2022. FACT SHEET: CHIPS and Science Act Will Lower Costs, Create Jobs, Strengthen Supply Chains, and Counter China. https: //www.whitehouse.gov/briefing-room/statements-releases/2022/08/09/factsheet-chips-and-science-act-will-lower-costs-create-jobs-strengthensupply-chains-and-counter-china/
- [50] 2022. Implementing Executive Order 13960 Promoting the Use of Trustworthy Artificial Intelligence in the Federal Government. https://www.nist.gov/ artificial-intelligence/EO13960 Last Modified: 2022-07-18T12:06-04:00.
- [51] 2022. Information Technology (IT) Annual Key Insights Report: Fiscal Year 2021 Successes and Accomplishments. Technical Report. IRS Office of the Chief Information Officer. https://www.irs.gov/pub/irs-pdf/p5453.pdf

- [52] 2022. Lessons Learned from Federal Use of Cloud Computing To Support Artificial Intelligence Research and Development. Technical Report. Machine Learning and Artificial Intelligence Subcommittee of the National Science and Technology Council. https://www.whitehouse.gov/wp-content/uploads/2022/07/07-2022-Lessons-Learned-Cloud-for-AI-July2022.pdf
- [53] 2022. The Networking & Information Technology R&D Program and the National Artificial Intelligence Initiative Office Supplement to the President's FY2023 Budget. Technical Report. Subcommittee on Networking and Information Technology Research and Development and the Machine Learning and Artificial Intelligence Subcommittee of the National Science and Technology Council. https://www. nitrd.gov/pubs/FY2023-NITRD-NAIIO-Supplement.pdf
- [54] 2022. Readout of the Tenth Meeting of the National Artificial Intelligence Research Resource (NAIRR) Task Force | OSTP. https://www.whitehouse. gov/ostp/news-updates/2022/10/27/readout-of-the-tenth-meeting-of-thenational-artificial-intelligence-research-resource-nairr-task-force/
- [55] 2023. AI Risk Management Framework: AI RMF (1.0). Technical Report NIST AI 100-1. National Institute of Standards and Technology, Gaithersburg, MD. 42 pages. https://doi.org/10.6028/NIST.AI.100-1
- [56] n.d.. About the Advisory Committee. https://www.ai.gov/naiac/
- [57] n.d.. Agency Inventories of AI Use Cases. https://www.ai.gov/ai-use-caseinventories/
- [58] n.d.. Agency Inventory of AI Use Cases. https://www.energy.gov/sites/default/ files/2022-07/DOE\_Agency\_Inventory\_of\_AI\_Use\_Cases.pdf
- [59] n.d.. AI Researchers Portal: Computing Resources. https://www.ai.gov/airesearchers-portal/computing-resources/
- [60] n.d.. AI Reviews & Algorithm Register | City of San José. https: //www.sanjoseca.gov/your-government/departments-offices/informationtechnology/digital-privacy/ai-reviews-algorithm-register
- [61] n.d.. AI Use Case Inventory Submission on Open Data. https://www.justice. gov/open/page/file/1517316/download
- [62] n.d. Algorithm Tips Resources and leads for investigating algorithms in society. http://algorithmtips.org/
- [63] n.d.. Artificial Intelligence (AI) Use Case Inventory. https://www.nsf.gov/data/ AI\_Inventory/
- [64] n.d.. Artificial Intelligence R&D InvestmentsFiscal Year 2018 Fiscal Year 2023. https://www.nitrd.gov/apps/itdashboard/ai-rd-investments/
- [65] n.d. Artificial Intelligence Use Case Inventory. http://www.dol.gov/agencies/ odg/ai-inventory
- [66] n.d.. The Cabinet. https://www.whitehouse.gov/administration/cabinet/
- [67] n.d.. The CFO Council. https://www.cfo.gov/
- [68] n.d.. Criterion: Evaluating Disparate Impact of Automated Decisions. https://certification.results4america.org/s/criterion/a0w2M000009ZuADQA0/ evaluating-disparate-impact-of-automated-decisions?language=en\_US
- [69] n.d.. Department of Energy Announces \$6.4 Million for Artificial Intelligence Research in High Energy Physics. https://www.energy.gov/science/articles/ department-energy-announces-64-million-artificial-intelligence-researchhigh
- [70] n.d.. Home | CBP Biometrics. https://biometrics.cbp.gov/
- [71] n.d.. Legislation and Executive Orders. https://www.ai.gov/legislation-andexecutive-orders/
- [72] n.d. NSF partnerships expand National AI Research Institutes to 40 states. https://beta.nsf.gov/news/nsf-partnerships-expand-national-ai-research
- [73] n.d.. The OECD Artificial Intelligence Policy Öbservatory. https://oecd.ai/en/ [74] n.d. OMB M-21-06 (Guidance for Regulation of Artificial Intelligence Applications). Technical Report. Department of Health and Human Services. https://www.hhs.gov/sites/default/files/department-of-health-andhuman-services-omb-m-21-06.pdf
- [75] n.d. Policies & Priorities: Executive Order (EO) 13960. https: //www.cio.gov/policies-and-priorities/Executive-Order-13960-AI-Use-Case-Inventories-Reference/
- [76] n.d.. Social Security Data Page. https://www.ssa.gov/data/
- [77] 117th Congress. 2022. James M. Inhofe National Defense Authorization Act for Fiscal Year 2023, Pub. L. 117–263, §7225, 136 Stat. 2395, 3671–72.
- [78] Alden F. Abbott. 1987. Case Studies on the Costs of Federal Statutory and Judicial Deadlines. Administrative Law Review 39, 4 (1987), 467–488. https: //heinonline.org/HOL/P?h=hein.journals/admin39&i=487
- [79] Julia Black. 2008. Forms and paradoxes of principles-based regulation. Capital Markets Law Journal 3, 4 (2008), 425–457.
- [80] Julia Black and Andrew Murray. 2019. Regulating AI and Machine Learning: Setting the Regulatory Agenda. European Journal of Law and Technology 10, 3 (2019), 1–21.
- [81] Alexander Bolton, Rachel Augustine Potter, and Sharece Thrower. 2016. Organizational Capacity, Regulatory Review, and the Limits of Political Control. *The Journal of Law, Economics, and Organization* 32, 2 (May 2016), 242–271. https://doi.org/10.1093/jleo/ewv025
- [82] John Braithwaite. 2002. Rules and Principles: A Theory of Legal Certainty. Australian Journal of Legal Philosophy 27 (2002), 47–82. https://doi.org/10.2139/ ssrn.329400

- [83] Evelyn Z. Brodkin. 2012. Reflections on Street-Level Bureaucracy: Past, Present and Future. Public Administration Review 72, 6 (2012), 940–949. https://doi.org/ 10.111/I.154O.621O.2O12.O2657.X. OCLC: ocn503596064.
- [84] Daniel Carpenter. 2004. Staff Resources Speed FDA Drug Review: A Critical Analysis of the Returns to Resources in Approval Regulation. *Journal of Health Politics, Policy and Law* 29, 3 (June 2004), 431–442. https://doi.org/10.1215/ 03616878-29-3-431
- [85] Corinne Cath and Fieke Jansen. 2023. Dutch Comfort: The limits of AI governance through municipal registers. *Techné: Research in Philosophy and Technol*ogy (2023).
- [86] Roger Clarke. 2019. Regulatory alternatives for AI. Computer Law & Security Review 35, 4 (Aug. 2019), 398-409. https://doi.org/10.1016/j.clsr.2019.04.008
- [87] Cary Coglianese. 2010. Public Availability of Agency Guidance Documents. Technical Report. Administrative Conference of the United States. https://www.acus.gov/sites/default/files/documents/Coglianese% 20Guidance%20Report%20to%20ACUS%2005.15.19%20-%20FINAL.pdf
- [88] Cary Coglianese. 2020. A framework for governmental use of machine learning. In Admin. Conf. United States.
- [89] Cary Coglianese. 2021. Administrative law in the automated state. Daedalus 150, 3 (2021), 104–120.
- [90] Cary Coglianese and David Lazer. 2003. Management-Based Regulation: Prescribing Private Management to Achieve Public Goals: Management-Based Regulation. Law & Society Review 37, 4 (Dec. 2003), 691–730. https://doi.org/ 10.1046/j.0023-9216.2003.03703001.x
- [91] Cary Coglianese and David Lehr. 2016. Regulating by robot: Administrative decision making in the machine-learning era. Geo. LJ 105 (2016), 1147.
- [92] Cary Coglianese and David Lehr. 2019. Transparency and algorithmic governance. Administrative Law Review 71, 1 (2019), 1–56.
- [93] Microsoft Corporation (Ed.). 2018. The future computed: artificial intelligence and its role in society. Microsoft, Redmond, Washington.
- [94] Matthew Dennis. 2022. Artificial Intelligence Strategic Plan, Fiscal Years 2023–2027: Draft Report for Comment. Technical Report NUREG-2261. U.S. Nuclear Regultory Commission. https://www.nrc.gov/docs/ML2217/ML22175A206. pdf
- [95] Grace Dille. 2022. President Biden Signs AI Workforce Training Act Into Law. https://www.meritalk.com/articles/president-biden-signs-ai-workforcetraining-act-into-law/
- [96] Colin S Diver. 1983. The Optimal Precision of Administrative Rules. Yale Law Journal 93, 1 (1983), 65–109.
- [97] Avinash Dixit. 2002. Incentives and Organizations in the Public Sector: An Interpretative Review. *The Journal of Human Resources* 37, 4 (2002), 696–727. https://doi.org/10.2307/3069614 Publisher: [University of Wisconsin Press, Board of Regents of the University of Wisconsin System].
- [98] Christian Djeffal, Markus B. Siewert, and Stefan Wurster. 2022. Role of the state and responsibility in governing artificial intelligence: a comparative analysis of AI strategies. *Journal of European Public Policy* 29, 11 (Nov. 2022), 1799–1821. https://doi.org/10.1080/13501763.2022.2094987
- [99] Richard F. Elmore. 1979. Backward Mapping: Implementation Research and Policy Decisions. *Political Science Quarterly* 94, 4 (1979), 601–616. https: //doi.org/10.2307/2149628
- [100] David Freeman Engstrom and Daniel E Ho. 2020. Algorithmic accountability in the administrative state. Yale J. on Reg. 37 (2020), 800.
- [101] David Freeman Engstrom and Daniel E Ho. 2021. Artificially intelligent government: A review and agenda. *Research Handbook on Big Data Law* (2021), 57-86.
- [102] David Freeman Engstrom, Daniel E. Ho, Catherine M. Sharkey, and Mariano-Florentino Cuéllar. 2020. Government by Algorithm: Artificial Intelligence in Federal Administrative Agencies. Technical Report. Administrative Conference of the United States. https://law.stanford.edu/wp-content/uploads/2020/02/ACUS-AI-Report.pdf
- [103] Thomas Ferretti. 2022. An Institutionalist Approach to AI Ethics: Justifying the Priority of Government Regulation over Self-Regulation. *Moral Philosophy* and Politics 9, 2 (Oct. 2022), 239–265. https://doi.org/10.1515/mopp-2020-0056 Publisher: De Gruyter.
- [104] Luciano Floridi. 2020. Artificial intelligence as a public service: Learning from Amsterdam and Helsinki. *Philosophy & Technology* 33, 4 (2020), 541–546.
- [105] Quint Forgey. 2021. Biden picks Samantha Power to lead USAID. https: //www.politico.com/news/2021/01/13/samantha-power-usaid-biden-458701
- [106] Andrew Freedman. 2022. Senate climate deal includes NOAA spending boost. https://www.axios.com/2022/08/03/senate-climate-deal-noaa-funding
- [107] William Funk. 2009. Public participation and transparency in administrative law-Three examples as an object lesson. Admin. L. Rev. 61 (2009), 171.
- [108] Urs Gasser and Virgilio Almeida. 2017. A Layered Model for AI Governance. IEEE Internet Computing 21, 6 (Nov. 2017), 58–62. https://doi.org/10.1109/mic. 2017.4180835

- [109] David K Hausman, Daniel E Ho, Mark S Krass, and Anne McDonough. 2022. Executive Control of Agency Adjudication: Capacity, Selection, and Precedential Rulemaking. The Journal of Law, Economics, and Organization (Oct. 2022), ewac012. https://doi.org/10.1093/ jleo/ewac012 \_\_eprint: https://academic.oup.com/jleo/advance-articlepdf/doi/10.1093/jleo/ewac012/46595665/ewac012.pdf.
- [110] Jory Heckman. 2021. HUD rolls out AI risk management platform to fight fraud in grant spending. https://federalnewsnetwork.com/artificialintelligence/2021/09/hud-rolls-out-ai-risk-management-platform-to-fightfraud-in-grant-spending/
- [111] Paul Henman. 2020. Improving public services using artificial intelligence: possibilities, pitfalls, governance. Asia Pacific Journal of Public Administration 42, 4 (2020), 209–221.
- [112] Daniel E. Ho, Jennifer King, Russell C. Wald, and Christopher Wan. 2021. Building a National AI Research Resource: A Blueprint for the National Research Cloud. Technical Report. Stanford Institute for Human-Centered Artificial Intelligence. https://hai.stanford.edu/sites/default/files/2022-01/HAI\_NRCR\_v17.pdf
- [113] Janet Holtzblatt and Alex Engler. 2022. Machine Learning and Tax Enforcement. Technical Report. Urban Institute & Brookings Institution Tax Policy Center. https://www.urban.org/sites/default/files/2022-06/Machine%20Learning% 20and%20Tax%20Enforcement.pdf
- [114] The White House. 2023. FACT SHEET: Biden-Harris Administration Announces New Actions to Promote Responsible AI Innovation that Protects Americans' Rights and Safety. https://www.whitehouse.gov/briefing-room/statementsreleases/2023/05/04/fact-sheet-biden-harris-administration-announces-newactions-to-promote-responsible-ai-innovation-that-protects-americansrights-and-safety/
- [115] John D. Huber and Nolan McCarty. 2004. Bureaucratic Capacity, Delegation, and Political Reform. *The American Political Science Review* 98, 3 (2004), 481–494. https://www.jstor.org/stable/4145342 Publisher: [American Political Science Association, Cambridge University Press].
- [116] Nigel Kingsman, Emre Kazim, Ali Chaudhry, Airlie Hilliard, Adriano Koshiyama, Roseline Polle, Giles Pavey, and Umar Mohammed. 2022. Public sector AI transparency standard: UK Government seeks to lead by example. *Discover Artificial Intelligence* 2, 1 (2022), 2.
- [117] Michael Kratsios. 2019. Why the US Needs a Strategy for AI. Wired (Feb. 2019). https://www.wired.com/story/a-national-strategy-for-ai/ Section: tags.
- [118] Ben S. Kuipers, Malcolm Higgs, Walter Kickert, Lars Tummers, Jolien Grandia, and Joris Van Der Voet. 2014. The Management of Change in Public Organizations: A Literature Review. *Public Administration* 92, 1 (March 2014), 1–20. https://doi.org/10.1111/padm.12040
- [119] Margaret B Kwoka. 2015. Foia, Inc. Duke LJ 65 (2015), 1361.
- [120] Christie Lawrence, Isaac Cui, and Daniel E. Ho. 2022. Implementation Challenges to Three Pillars of America's AI Strategy. https://reglab.stanford.edu/publications/implementation-challenges-tothree-pillars-of-americas-ai-strategy/
- [121] Soo-Young Lee and Andrew B. Whitford. 2013. Assessing the Effects of Organizational Resources on Public Agency Performance: Evidence from the US Federal Government. Journal of Public Administration Research and Theory: J-PART 23, 3 (2013), 687–712. https://www.jstor.org/stable/24484865 Publisher: [Oxford University Press, Journal of Public Administration Research and Theory, Inc., Public Management Research Association].
- [122] Michael Lipsky. 1971. Street-Level Bureaucracy and the Analysis of Urban Reform. Urban Affairs Quarterly 6, 4 (June 1971), 391–409. https://doi.org/10. 1177/107808747100600401
- [123] Giandomenico Majone. 1997. From the Positive to the Regulatory State: Causes and Consequences of Changes in the Mode of Governance. *Journal of Public Policy* 17, 2 (May 1997), 139–167. https://doi.org/10.1017/S0143814X00003524
- [124] Colin D. Moore. 2015. Innovation without Reputation: How Bureaucrats Saved the Veterans' Health Care System. *Perspectives on Politics* 13, 2 (June 2015), 327–344. https://doi.org/10.1017/S1537592715000067
- [125] National Security Commission on Artificial Intelligence. 2021. The Honorable Gina Raimondo, U.S. Secretary of Commerce - 04 - NSCAI Global Emerging Tech Summit. https://www.youtube.com/watch?v=9j0tOfMOnwA
- [126] Dave Nyczepir. 2022. Education Department replacing grants management system to handle awards increase. *FedScoop* (April 2022). https://fedscoop.com/ education-department-replaces-grants-system/
- [127] Teresa Dale Pupillo. 1993. The Changing Weather Forecast: Government in the Sunshine in the 1990's-An Analysis of State Sunshine Laws. Wash. ULQ 71 (1993), 1165.
- [128] Dillon Reisman, Jason Schultz, Kate Crawford, and Meredith Whittaker. 2018. Algorithmic Impact Assessments: A Practical Framework for Public Agency. AI Now (2018).
- [129] Daniel S. Schiff. 2023. Looking through a policy window with tinted glasses: Setting the agenda for U.S. AI policy. *Review of Policy Research* n/a, n/a (2023). https://doi.org/10.1111/ropr.12535 \_eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/ropr.12535.

- [130] Jennifer L Selin and David E Lewis. 2018. Sourcebook of United States Executive Agencies. Technical Report. Administrative Conference of the United States. https://www.acus.gov/sites/default/files/documents/ACUS%20Sourcebook% 200f%20Executive%20Agenices%202d%20ed.%20508%20Compliant.pdf
- [131] Jennifer L. Selin and David E. Lewis. 2018. Sourcebook of United States Executive Agencies (Second Edition): Sourcebook Data. https://www.acus.gov/appendix/ sourcebook-data
- [132] M.E.W.M. Silkens, J. Ross, M. Hall, H. Scarbrough, and A. Rockall. 2023. The time is now: making the case for a UK registry of deployment of radiology artificial intelligence applications. *Clinical Radiology* 78, 2 (2023), 107–114. https: //doi.org/10.1016/j.crad.2022.09.132 Special Issue Section: Artificial Intelligence and Machine Learning.
- [133] Richard Stirling, Pasquarelli, and Eleanor Shearer. 2020. Shared: Readiness Framework Online Version.pdf. Technical Report. Oxford Insights. https://drive.google.com/file/d/1hiTjUDITecHi09y-AystweXWF7VmRsj6/view?usp=sharing&usp=embed\_facebook
- [134] Hugo Teufel III. 2008. Memorandum Number 2008-02, Privacy Policy Guidance Memorandum: DHS Policy Regarding Privacy Impact Assessments. https://www.dhs.gov/sites/default/files/publications/privacy\_ policyguide\_2008-02\_0.pdf
- [135] Ian R. Turner. 2020. Policy Durability, Agency Capacity, and Executive Unilateralism. Presidential Studies Quarterly 50, 1 (2020), 40-62. https://doi.org/10. 1111/psq.12633
- [136] Inga Ulnicane, William Knight, Tonii Leach, Bernd Carsten Stahl, and Winter-Gladys Wanjiku. 2020. Framing governance for a contested emerging technology:insights from AI policy. Pol'y Soc'y 40 (2020), 158–177. Issue 2.
- [137] James L. Vann. 2004. Resistance to Change and the Language of Public Organizations: A Look at "Clashing Grammars" in Large-Scale Information Technology Projects. *Public Organization Review* 4, 1 (March 2004), 47–73. https://doi.org/10.1023/B:PORJ.0000015651.06417.e1
- [138] Russell T. Vought. 2020. M-21-06, Memorandum for the Heads of Executive Departments and Agencies: Guidance for Regulation of Artificial Intelligence Applications. https://www.whitehouse.gov/wp-content/uploads/2020/11/M-21-06.pdf
- [139] Kathleen Walch. 2020. How The Federal Government's AI Center Of Excellence Is Impacting Government-Wide Adoption Of AI. https://www.forbes.com/sites/cognitiveworld/2020/08/08/how-the-federalgovernments-ai-center-of-excellence-is-impacting-government-wideadoption-of-ai/ Section: AI.
- [140] Todd R. Weiss. 2020. COVID-19: How AI is helping to streamline SBA payroll loans to small businesses. https://www.techrepublic.com/article/covid-19-howai-is-helping-to-streamline-sba-payroll-loans-to-small-businesses/
- [141] James Q. Wilson. 2000. Bureaucracy: What Government Agencies Do and Why They Do It (new ed.). Basic Books, New York, NY.
- [142] Titus Wu. 2023. California Seeks to Be First to Regulate Business Use of AI. https://news.bloomberglaw.com/in-house-counsel/california-seeks-tobe-first-to-regulate-business-use-of-ai
- [143] Daniel Zhang, Christie Lawrence, Michael Sellitto, Russell Wald, Marietje Schaake, Daniel E Ho, Russ Altman, and Andrew Grotto. 2022. Enhancing International Cooperation in AI Research: The Case for a Multilateral AI Research Institute. Technical Report. Stanford Institute for Human-Centered Artificial Intelligence. https://hai.stanford.edu/sites/default/files/2022-05/HAI%20Policy%20White%20Paper%20-%20Enhancing%20International% 20Cooperation%20in%20AI%20Research.pdf
- [144] Daniel Zhang, Nestor Maslej, Erik Brynjolfsson, John Etchemendy, Terah Lyons, James Manyika, Helen Ngo, Juan Carlos Niebles, Michael Sellitto, Ellie Sakhaee, Yoav Shoham, Jack Clark, and Raymond Perrault. 2022. The AI Index 2022 Annual Report. Technical Report. Stanford Institute for Human-Centered Artificial Intelligence. https://aiindex.stanford.edu/wp-content/uploads/2022/03/2022-AI-Index-Report\_Master.pdf

### A IMPLEMENTATION OF LEGAL REQUIREMENTS

### A.1 Methodology

To assess the implementation status of the AI Leadership Order, the Trustworthy AI Order, and the AI in Government Act, we first identified all line-level actions that these documents mandate (e.g., instructions that a federal entity "shall budget," "shall consider," "shall review," "shall publish"). For each requirement, the following information was compiled in a tracker (see Appendix E): (1) the relevant portion of the executive order or legislation, (2) the government stakeholder responsible for its implementation, (3) a summary of the mandated outcome or deliverable, (4) the mandated deadline, if any, and (5) the "type" of requirement (see below paragraph), and (6) the status of implementation. The first four items were drawn from the text of the executive order or legislation itself, while the status of implementation was drawn from publicly available information, as of November 23, 2022. Where possible, we provide additional details about the implementation of the requirement and URL links to relevant documents. Therefore this represents the publicly verifiable status and may not capture activities executed without public disclosure (either with the intent to protect sensitive or classified information or simply because the federal entities did not prioritize or have an appropriate avenue for disclosing the activity).

As noted above, the requirements were split into three categories. This facilitated assessment of implementation by the responsible federal government entity (see tables in Appendix E.1). The categories were:

- (1) *Time-boxed requirements* mandated a federal entity, or entities, to produce a document or achieve an outcome by a specified date (e.g., "shall develop" a report within 90 days of the date of the executive order).
- (2) Open-ended requirements mandated the production of a document/deliverable or achievement of an outcome without specifying a deadline.
- (3) Ongoing requirements were open-ended mandates to agencies that often did not require the production of a specific document/deliverable or achievement of a concrete outcome (e.g., agencies "shall pursue" an objective, "shall consider" actions, "shall identify opportunities," "shall provide" expertise, etc.). These ongoing requirements also did not have a deadline. This also includes outcomes that were part of an annual process without a specified date (e.g., the AI Leadership Order's requirements in Section 4(b)-(b)(1) that agencies prioritize AI R&D and "communicate plans for achieving this prioritization to the OMB Director and the OSTP Director").

Although assessing implementation of the time-boxed requirements was often straightforward, compliance with a significant percentage of mandated actions was not known or hard to determine, either because the mandate required ongoing compliance without producing a specific milestone, the mandated action did not require public disclosure of its completion or progress toward its completion, or both. Under the assumption that federal entities had taken necessary steps, or at least made good faith efforts, to meet their legal and statutory requirements, ambiguity was resolved in favor of the federal entities. Therefore, the researchers applied the following rules for determining implementation status: implemented (or indications of implementation), not implemented (or indications that the requirement was not implemented), and not known.

- *Implemented*: Time-boxed requirements were marked as successfully implemented where the mandated outcome was achieved, even if achieved after the mandated deadline. Openended requirements and ongoing requirements without a defined deliverable were coded green if public information strongly supported the conclusion that federal entities were implementing the requirement.
- Not Implemented: Time-boxed requirements were marked as not implemented if there was no public information, as of November 23, 2022, confirming their implementation by the mandated deadline. Requirements were coded red if public information strongly suggested that they had not been implemented by federal entities. The latter, for instance, occurred for the AI Leadership Order's requirement for a National Security Presidential Memorandum.
- Not Known: Implementation of time-boxed requirements and open-ended requirements was marked as not known where public reporting was nonexistent, often because public reporting was not mandated, or did not clearly indicate the status of implementation. Similarly, the implementation of ongoing requirements was marked as not known because there was no mandated reporting and often no mandated outcome for the researchers to publicly verify.

### A.2 Summary of Findings

A summary of the findings for each document is provided in Tables 1, 4, 5, and 6. The detailed methodology is provided in Appendix A.1, but it is important to highlight upfront that methodological constraints may result in our findings underestimating implementation and overestimating requirements that remain outstanding. Although best efforts were taken to properly identify all relevant documents or notices of actions, the researchers could only rely on federal entities' public disclosures, which may not capture all relevant actions taken by the federal government to achieve the mandates.

- AI Executive Order: Only 39 percent, or nine of the order's 23 requirements, were implemented. Given a dearth of publicly available information about many of the requirements, the implementation status for a majority of the requirements was not known (57 percent). Requirements with a specified deadline had a higher rate of implementation (45 percent) than requirements without a deadline (0 percent) or open-ended requirements without a concrete deliverable (40 percent). Critically, the requirement for agencies to publish AI Plans to achieve consistency with OMB guidance on regulating AI was not fulfilled. The implementation of these Agency AI Plans is discussed in Section 6 and Appendix B.2.
- Trustworthy AI Order: Implementation was even lower for the Trustworthy AI Order, with only 13 percent, or two of the

requirements, implemented.<sup>36</sup> Similar to the AI Leadership Order, implementation for a majority of the requirements (54 percent) could not be conclusively determined. Two of the requirements, or 13 percent, have not been implemented, including the requirement for agencies to prepare and publish AI use case inventories. The implementation of these Agency AI use case inventories is discussed in Section 7 and Appendix C.2.

• AI in Government Act of 2020: Compared to the executive orders, the percentage of requirements that were not implemented was much higher at 67 percent, or four of the six requirements. The only requirement implemented was to establish an AI Center of Excellence within GSA, while the progress that GSA has made on achieving the Center of Excellence's duties is unknown.

### **B** IMPLEMENTATION OF AGENCY AI PLANS

### **B.1** Methodology and Background

*B.1.1* Background on AI Leadership Order's "Agency AI Plan" Requirement. As discussed in Section 6, a significant focus of the AI Leadership Order was addressing concerns about regulatory gaps and hurdles to AI development and deployment. As such, the executive order mandated:

- The White's Office of Management and Budget (OMB) to issue a guidance memorandum to agencies, after publishing a draft guidance for public comment, within 180 days of the EO (approximately August 2019). Sections 6(a)-(b).
- The heads of "implementing agencies" with regulatory authorities to develop a plan to "achieve consistency" with the OMB memorandum within 180 days of OMB issuing the memorandum. Section 6(c).

OMB fulfilled its requirement over a year overdue, publishing a draft memorandum on January 1, 2020,[27] and issuing its final memorandum on November 17, 2020.[138] OMB M-21-06, Memorandum for the Heads of Executive Departments and Agencies on Guidance for Regulation of Artificial Intelligence Application (referred to as the "OMB AI Regulation Memo" for ease of understanding), provided guidance for agencies on regulatory and nonregulatory approaches to AI. Critically, it noted that "government use of AI" was outside of the scope of the memorandum.

The OMB AI Regulation Memo also provided guidance for the Agency AI Plans. Specifically, it stated:

The agency plan must identify any statutory authorities specifically governing agency regulation of AI applications, as well as collections of AI-related information from regulated entities. For these collections, agencies should describe any statutory restrictions on the collection or sharing of information (e.g., confidential business information, personally identifiable information, protected health information, law enforcement information, and classified or other national security information). The agency plan must also report on the outcomes of stakeholder engagements that identify existing regulatory barriers to AI applications and high-priority AI applications that are within an agency's regulatory authorities. OMB also requests agencies to list and describe any planned or considered regulatory actions on AI.

Furthermore, the memorandum included specific instructions for how agencies must submit and publish their plans:

Agency plans are due on May 17, 2021, and should be submitted to OIRA at the following email address: Alplans@omb.eop.gov. To inform the public of each agency's planned and implemented activities, agency plans must be posted on, or be accessed from (through a URL redirect), the following domain on the agency's website: www.[agencyname].gov/guidance.

The May 2021 deadline adhered to the AI Leadership Order's requirement that the plans be completed and submitted within 180 days of the OMB AI Regulation Memo's issuance.

The OMB AI Regulation Memo did not provide guidance on which agencies were subject to the executive order's requirements. The AI Leadership Order stated that the requirement applied to "implementing agencies that also have regulatory authorities." "Implementing agencies" were defined in Section 3 of the AI Leadership Order as "agencies that conduct foundational AI R&D, develop and deploy applications of AI technologies, provide educational grants, and regulate and provide guidance for applications of AI technologies, as determined by the co-chairs of the NSTC Select Committee." This set is potentially quite broad, especially as regulation of applications of AI would include many incumbent regulatory regimes (e.g., approval of medical devices by the Food and Drug Administration, discrimination of employment policies by the Equal Employment Opportunity Commission). However, the NSTC Select Committee on AI did not publish a list of agencies it determined were "implementing agencies," nor did the OMB AI Regulation Memo provide any additional insight. Although the OMB AI Regulation Memo directed itself to "heads of all Executive Branch departments and agencies, including independent regulatory agencies," neither the memorandum nor the executive order defined "regulatory authorities," a potentially expansive term subsuming most administrative agencies, or delineated which agencies had regulatory authorities.

*B.1.2 Methodology for Assessing Implementation.* To identify relevant agencies, we first searched online for a list of agencies deemed to be "implementing agencies" by the co-chairs of the NSTC Select Committee on AI. As this list was not publicly available, we instead focused on Cabinet-level departments and agencies and the 19 agencies deemed "independent regulatory agencies" under 44

<sup>&</sup>lt;sup>36</sup>There were 17 requirements, but one requirement was excluded from the overall calculations because its deadline has not yet passed (i.e., the rate of implementation assumed 16 instead of 17 requirements).

### Table 4: Summary of Federal Entities' Implementation of Requirements in AI Leadership Order (EO 13,859)

	Relevant Sections	Implemented	Unknown Implemen- tation	Not Implemented
11 Time-Boxed Require- ments	5(a)(i) <sup><i>a</i></sup> , 5(a)(ii), 5(a)(iii), 5(c), 6(a)-(b), 6(c), 6(d), 7(b), 8(a)- (b), 8(c)	45%	45%	9%
		5 (5(a)(i) <sup><i>a</i></sup> , 5(c), 6(a)-(b), 6(d))	5 (5(a)(ii), 5(a)(iii), 7(b), 8(a)-(b), 8(c))	$1 (6(c)^{b}.)$
2 Open-Ended Require- ments	5(a), 5(a)(iv)	0%	100%	0%
		_	2 (5(a), 5(a)(iv))	_
10 Ongoing Require- ments	$\begin{array}{llllllllllllllllllllllllllllllllllll$	40%	60%	0%
		4 (4(a), 4(b)-(b)(i), 4(b)(ii), 5(b))	6 (2(a)-(e), 4(c), 5(a)(v), 5(d), 7(a)(i)-(ii), 7(c))	-

 $^a$  5(a)(i) has two time-boxed requirements.  $^b \rm See$  Appendix E.2

### Table 5: Summary of Federal Entities' Implementation of Requirements in Trustworthy AI Order (EO 13,960)

	Relevant Sections	Implemented	Unknown Implemen- tation	Not Implemented
12 Time-Boxed Require- ments <sup>a</sup>	4(b), 5(a), 5(b), 5(c)(i), 5(c)(ii), 5(d), 5(e), 6 <sup>b</sup> , 7(a), 7(b), 7(c), 8(c)	17%	58%	17%
		2 (5(a), 7(a))	7 (5(b), 5(c)(i), 5(d), 6 <sup>b</sup> , 7(b), 7(c), 8(c))	2 (4(b), 5(e) <sup>c</sup> )
1 Open-Ended Require- ment	5(c)	0%	100%	0%
		_	1 (5(c))	_
4 Ongoing Requirements	2(b), 4(a), 4(c), 6 <sup>b</sup>	0%	100%	0%
		_	4 (2(b), 4(a), 4(c), 6 <sup>b</sup> )	_

<sup>a</sup>Requirement in section 5(c)(ii) has not been implemented, but the deadline for implementation has not yet passed, so it is not classified as implemented, not implemented, or not known. Therefore the percentages for the 12 time-boxed requirements do not equal 100 percent.

<sup>b</sup>6 has one time-boxed requirement and one ongoing requirement. See Appendix E.1

<sup>c</sup>See Section 7, Table 3, and Appendix C.

U.S.C. § 3502(5).<sup>37</sup> We also included the U.S. Agency for International Development (USAID), as it was the only agency represented at the National Security Council[105] that was not already included as a Cabinet-level agency or as an independent regulatory agency. The reason for including each agency is identified in the full tracker

in Appendix E.2. It is possible that this list is overinclusive or underinclusive of the agencies that were actually required to establish and publish an Agency AI Plan to achieve consistency with the OMB AI Regulation Memo. We also inquired with a member of the Select Committee and did not receive an answer on what agencies are included.

The intended purpose of the OMB AI Regulation Memo's requirement that the Agency AI Plans should be available on the respective agency website's page on guidance was to increase transparency and "inform the public." Therefore, identifying the plans should be intuitive to the public and should not require significant expenditure of time. To simulate how an individual might seek to access the plan, we implemented four simple approaches to finding

<sup>&</sup>lt;sup>37</sup>The current Cabinet includes the heads of the 15 executive departments (the Secretaries of Agriculture, Commerce, Defense, Education, Energy, Health and Human Services, Homeland Security, Housing and Urban Development, Interior, Labor, State, Transportation, Treasury, and Veterans Affairs, and the Attorney General), the White House Chief of Staff, the U.S. Ambassador to the United Nations, the Director of National Intelligence, and the U.S. Trade Representative, as well as the heads of the Environmental Protection Agency, Office of Management and Budget, Council of Economic Advisers, Office of Science and Technology Policy, and Small Business Administration.[66] We excluded the White House Chief of Staff, U.S. Ambassador to the U.N., and Council of Economic Advisors because they do not have rule-making or regulatory authority.[130]

	Relevant Sections	Implemented	Unknown Implemen- tation	Not Implemented
4 Time-Boxed Require- ments	104(a)-(b) & (d), 104(c), 105(a), 105(b)	0%	0%	100%
		_	-	4 (104(a)-(b) & (d), 104(c), 105(a), 105(b))
1 Open-Ended Require- ment	103	100%	0%	0%
		1 (103)	_	_
1 Ongoing Requirement	103	0%	100%	0%
		_	1 (103)	_

Table 6: Summary of Federal Entities' Implementation of Requirements in AI in Government Act of 2020

it. For each, the researchers noted whether the approach yielded a positive identification of an Agency AI Plan in the correspondingly titled columns in E.2.

- Dedicated Agency URL: Visiting the link under which the OMB AI Regulation Memo expressly requires the Agency AI Plan to be posted: [agency\_name].gov/guidance. We noted first if the agency had a dedicated guidance webpage. If it did, we searched "response artificial intelligence OMB M-21-06" (as "OMB" and "M-21-06" are expressly noted in the template response). If it did not, we marked "no" for this method.
- Web Search: We searched online (using Google) "[agency name] response artificial intelligence OMB M-21-06". If the agency's full name did not return results, we searched with the agency's acronym (e.g., HHS for the Department of Health and Human Services), where applicable.
- Search Within Agency Website: Searching within an agency's website: "response artificial intelligence OMB M-21-06." If (as noted above) an agency lacked its own website, we searched on its parent agency's website with its name included, e.g., "[agency name] response artificial intelligence OMB M-21-06". If the search engine returned an implausibly large number of results (e.g., on the order of 10,000), phrases would be placed in quotation marks (e.g., "artificial intelligence," "use case," and "M-21-06").
- **AI.gov**: Searching the publication library on AI.gov (the website for the National AI Initiative) for the agency's name (or acronym) and "response artificial intelligence OMB M-21-06". We also looked at all documents published by that agency and included in the publication library as there was a small number of documents per agency, if any, in the publication library.

If an Agency AI Plan was identified using any of these four methods, as of November 23, 2022, the researchers marked "yes" in the "Agency Plan" column (Appendix E.2) and provided the web link to the plan in the "URL" column. If the Agency AI Plan was not identified using any of the four methods, the researchers marked "no" for the presence of an "Agency Plan."

### **B.2 Summary of Findings**

The agencies with an Agency AI Plan are the Departments of Energy, HHS, and VA, the EPA, and USAID (see Tables 2 and 7).

The agencies without AI Plans are the Departments of Agriculture (USDA), Commerce, Defense, Education, Homeland Security, Housing and Urban Development, Interior, Justice, Labor, State, Transportation, Treasury, and the Board of Governors of the Federal Reserve System (FED), Commodity Futures Trading Commission (CFTC), Consumer Financial Protection Bureau (CFPB), Consumer Product Safety Commission (CPSC), Federal Communications Commission (FCC), Federal Deposit Insurance Corporation (FDIC), Federal Energy Regulatory Commission (FERC), Federal Housing Finance Agency (FHFA), Federal Maritime Commission (FMC), Federal Trade Commission (FTC), Mine Enforcement Safety and Health Review Commission (FMSHRC), National Labor Relations Board (NLRB), Nuclear Regulatory Commission (NRC), Occupational Safety and Health Review Commission (OSHRC), Office of Financial Research (OFR), Office of Management and Budget (OMB), Office of Science and Technology Policy (OSTP), Office of the Comptroller of the Currency (OCC), Office of the Director of National Intelligence (ODNI), Office of the U.S. Trade Representative (USTR), Postal Regulatory Commission (PRC), Securities and Exchange Commission (SEC), Small Business Administration (SBA), and the Surface Transportation Board (STB).

Four agencies published AI-related strategic plans, including some that noted the AI Leadership Order, but these plans provided far less than the detailed required. The DHS's S&T AI and ML Strategic Plan [38], the VA's AI Strategy [30], and the Department of State's Enterprise Data Strategy [33] mention the AI Leadership Order and identify AI priorities but provide less detail on regulation than required under the AI Leadership Order. The Nuclear Regulatory Commission published an AI Strategic Plan in June 2022 [94], but it similarly does not provide enough detail to classify as an Agency AI Plan consistent with the OMB AI Regulation Memo.

Table 7: Summary	of Agencies y	with Agency	AI Plans
rable 7. Summary	of figure to v	when Ageney	111 I Ians

Agency	Overview of the Substance in Agency's AI Plan
Department of Energy (DOE)	• Input "none" for each of the five questions
	- mpar none for each of the five questions
Department of Health and Human Services (HHS)	<ul> <li>11 statutes that authorized HHS to regulate AI applications, even noting that two of the statutes do not directly mention AI but might provide indirect authority to regulate AI as it relates to health data or health technology</li> <li>32 active collections of AI-related information, 30 of which were approved by OMB pursuant to the Paperwork Reduction Act and two that were exempted from OMB clearance as they are "general requests"</li> <li>12 AI use case priorities, 7 were AI applications in the private sector that were under its regulatory authorities (e.g., AI algorithm for wrist fracture reduction), 4 were opportunities for HHS to "shape the development and production of AI in the private sector," such as creating and improving relevant datasets, and 1 (predicting risk of adult maltreatment) was an internal AI tool that could be adopted by the private sector</li> <li>10 AI regulatory barriers (e.g., data silos, intellectual property, concerns about HIPAA and data sharing)</li> <li>4 planned regulatory actions concerning AI applications (e.g., imposing clinical holds on medical devices)</li> </ul>
Department of Veterans Affairs (DVA)	<ul> <li>No statutory authorities directing or authorizing agency regulation of AI</li> <li>No active collections of AI-related information</li> <li>14 AI use case priorities (e.g., identifying risk factors for diseases or suicide risk, AI that triages incoming medical evidence like images or lab results)</li> <li>3 AI regulatory barriers, which were all specific regulations (HIPAA, Electronic Communications Privacy Act, Privacy Act of 1974 amended as 5 U.S.C. 552a)</li> <li>No planned regulatory actions concerning AI</li> </ul>
Environmental Protection Agency (EPA)	<ul> <li>No statutory authorities directing or authorizing agency regulation of AI</li> <li>No active collections of AI-related information</li> <li>No AI use cases in private sector within regulatory authority, but a handful of AI use cases identified as of interest for achieving EPA's goals</li> <li>No AI regulatory barriers identified</li> <li>No planned regulatory actions concerning AI, but noted EPA began working on AI strategies like technical architectures and eventually higher-level principles</li> </ul>
U.S. Agency for International Develop- ment (USAID)	<ul> <li>No statutory authorities directing or authorizing agency regulation of AI</li> <li>No active collections of AI-related information</li> <li>No AI use cases in private sector within regulatory authority</li> <li>No AI regulatory barriers identified</li> <li>No planned regulatory actions concerning AI</li> </ul>

# C IMPLEMENTATION OF AI USE CASE INVENTORIES

### C.1 Methodology

*C.1.1* Background on the Trustworthy AI Order's AI Use Case Inventory Requirement. Agencies were to prepare their inventories within 180 days of the Federal Chief Information Officers Council

(CIO Council) providing guidance to the agencies (which occurred in fall 2021 [75]) and annually thereafter. The CIO guidance instructed agencies to report use case inventories using a provided Excel template by March 22, 2022.

*Responsible Agencies*: Agencies that must comply were defined by the Trustworthy AI Order[25] in Section 8 as "all agencies described

in section 3502, subsection (1), of title 44, United States Code, except for the agencies described in section 3502, subsection (5), of title 44." The Department of Defense and "those agencies and agency components with functions that lie wholly within the Intelligence Community" were also exempted.

Scope: The Trustworthy AI Order used the definition of AI "set forth in section 238(g) of the National Defense Authorization Act for Fiscal Year 2019 as a reference point."38 The order further clarified in Section 9 that it applied to "both existing and new uses of AI; both standalone AI and AI embedded within other systems or applications; AI developed both by the agency or by third parties on behalf of agencies for the fulfillment of specific agency missions, including relevant data inputs used to train AI and outputs used in support of decision making; and agencies' procurement of AI applications." However, the order excluded some AI uses from the AI inventory requirement, including "AI used in defense or national security systems (as defined in 44 U.S.C. 3552(b)(6) or as determined by the agency)," "AI embedded within common commercial products, such as word processors or map navigation systems," and "AI research and development (R&D) activities." The CIO's Example AI Use Case Inventory Scenarios provides additional guidance.[75]

*Submission and Publication*: Given the timing of the CIO's issuance of the guidance, the CIO guidance [75] instructed:

By March 22, 2022, Agencies shall use the provided Excel workbook, 'Agency AI Use Case Inventory,' to compile their AI use cases and upload one file per agency to the MAX site at: Agency AI Inventory Instructions and Submission - E-Government Community - MAX Federal Community.

This guidance adhered to the Trustworthy AI Order, which mandated that agencies share their inventories with other agencies within 60 days of completing them and then make their inventories publicly available within 120 days of completing their inventories.

*C.1.2 Methodology for Assessing Implementation.* To identify relevant agencies, we looked to the ACUS Sourcebook of U.S. Executive Agencies ("ACUS Sourcebook")[130] and included all 278 agencies and sub-agencies identified in the Sourcebook data spreadsheet.[131] Given the Trustworthy AI Order's explicit exclusions, we removed agencies within the Department of Defense, agencies and sub-agencies within the intelligence community as defined by 50 U.S.C. § 3003(4), and the 19 independent regulatory agencies defined in 44 U.S.C. § 3502(5).<sup>39</sup> We further made individualized adjustments for agencies that are now defunct or are administered under different names.<sup>40</sup> This produced a total of 220 agencies.

Similar to the Agency AI Plans, identifying AI inventories should be intuitive to the public and should not require significant expenditure of time. We implemented four approaches:

- (1) Dedicated Agency URL: We visited the relevant website as provided by the CIO 2021 Guidance for Creating Agency Inventories of AI Use Case [75]: "[agency\_name].gov/data/-AI\_Inventory." If the relevant agency's webpage did not lead to an AI inventory, or if the agency did not have a URL of that form, we recorded "Dedicated Agency URL" as "No."
- (2) **Web Search**: We searched online (using Google) "[agency name] artificial intelligence use case inventory." If the agency's full name did not return results, we also searched using the agency's acronym or more common name (e.g., DHS or Farmer Mac).
- (3) **Search Within Agency Website**: We searched within an agency's website (i.e., using its internal search engine): "artificial intelligence use case inventory." If (as noted above) an agency lacked its own website, we searched on its parent agency's website with its name included, e.g., "[agency name] artificial intelligence use case inventory." If the search engine returned an implausibly large number of results (e.g., on the order of 10,000), phrases would be placed in quotation marks.
- (4) **AI.gov**: We searched AI.gov's (website for the National AI Initiative) tracker for agency AI use case inventories.

Multiple measures were employed to measure implementation. The measurements varied along two major dimensions:

- (1) Agencies considered: We measured implementation rates by considering different subsets of agencies—specifically, we employed three agency groupings: all relevant agencies, large agencies, and agencies with a known AI use case. Appendix E.3 includes the list of all 220 agencies and classifies which agencies are large and have a known AI use case.
  - (a) All relevant agencies considers all 220 agencies identified using the methodology described above. This approach does not consider agency size or likelihood of the agency employing AI.
  - (b) Large agencies considers 125 "large" agencies. To identify this subset, we benchmarked against the 2020 "Government by Algorithm: Artificial Intelligence in Federal Administrative Agencies" report submitted to ACUS ("ACUS AI Report").[102] The ACUS AI Report narrowed the agencies listed in the 2018 ACUS Sourcebook by (1) including only agencies with more than 400 employees; and (2) removing active military and intelligence-related agencies. The ACUS AI Report therefore identifies 142 "large" agencies. For this tracker, the 142 agencies had to be further

<sup>&</sup>lt;sup>38</sup>The order noted in Section 9(a) that the evolution of AI use in the federal government necessitates that "OMB guidance developed or revised pursuant to section 4 of this order shall include such definitions as are necessary to ensure the application of the Principles in this order to appropriate use cases."

<sup>&</sup>lt;sup>39</sup>One of the named independent regulatory agencies within 44 U.S.C. § 3502(5) is the Interstate Commerce Commission, which is now defunct. We excluded its successor, the Surface Transportation Board.

<sup>&</sup>lt;sup>40</sup>Specifically, we excluded: (1) the National Association of Registered Agents and Brokers (NARAB), which was statutorily created in 2015 but not implemented; (2) the Federal Supplementary Medical Insurance Trust Fund and the Federal Hospital Insurance Trust Fund, both of which are administered by the Medicare Board of Trustees, which is what we have included in the Tracker; (3) the Office of Healthy Homes and Lead Hazard Control, which is currently known as the Office of Lead Hazard Control and Healthy Homes (included in the Tracker); (4) the Grain Inspection, Packers, and Stockyards Administration, whose functions are now housed in the

Agricultural Marketing Service; (5) the Northern Great Plains Regional Authority, which is now defunct; (6) the Economic and Statistics Administration, which no longer exists; and (7) the Internal Revenue Service Oversight Board, which has been suspended. We further added the Executive Office of the President as a parent agency. Though it is probably best regarded as not an "agency" ([130, p. 19]). Notably, we did not exclude three agencies in the Department of Agriculture listed by the ACUS Sourcebook—the Rural Business-Cooperative Service, the Rural Housing Service, and the Rural Utilities Service—that seem to be child-agencies of a USDA sub-agency known as "Rural Development." To our knowledge, these three are the only examples of sub-sub-agencies featured in our analysis.

narrowed by removing the independent regulatory agencies within the meaning of 44 U.S.C. § 3502(5) and the now-defunct agencies.<sup>41</sup> The result is a total of 125 agencies.<sup>42</sup>

- (c) Agencies with a known AI use case considers 49 agencies with a non-zero number of AI use cases identified by the ACUS AI Report team.<sup>43</sup> The ACUS AI Report identified through "an agency-by-agency, web-based search protocol, augmented by a range of third-party sources" any use case where an agency "had considered using or had already deployed AI/ML technology to carry out a core function," discounting instances "where agencies demonstrated no intent to operationalize a given tool," such as "a pure research paper using AI/ML." Because the ACUS team focused on whether the agency was deploying AI for a "core function," identifying an AI use case is a decent proxy for presuming that that agency ought to report some inventory pursuant to the Trustworthy AI Order. If the agency did not have an inventory but it did have a non-zero number of use cases, we classify that agency as not having implemented the requirement.44
- (2) **Organizational level**: We calculate the compliance and noncompliance rate at both the *individual/sub-agency* and *parent* level. Appendix E.3 identifies the parent agency and its sub-agencies.
  - (a) At the *individual/sub-agency level*, we disaggregate all sub-agencies from their parent agency. Because nearly all inventories were published by the parent-level agency,<sup>45</sup> we denoted a sub-agency as having published an inventory if its use cases are described and assigned to that sub-agency within the parent agency's inventory.<sup>46</sup> We

calculated the implementation rate by dividing the number of sub-agencies and parent-level agencies with a use case inventory by the total number of agencies for that measure (i.e., 220 for "all relevant agencies," 125 for "large agencies," and 49 for "large agencies with known AI uses").

- (i) For example, the Department of Justice, which has 14 sub-agencies, published an AI use case inventory that included use cases from two of its sub-agencies that were in our set of agencies (Drug Enforcement Administration and Federal Bureau of Investigation).<sup>47</sup> Because DOJ, DEA, and FBI have or are listed in a use case inventory, they are marked as having implemented an inventory, while the remaining 12 sub-agencies not included in DOJ's AI use case inventory are marked otherwise; the non-implementation rate is thus 80% (12/15).
- (b) At the *parent* level, we bundle all sub-agencies' use cases into the parent-level agency. There are 78 parent-level agencies.
  - (i) For example, instead of counting the Department of Commerce and all of its sub-agencies, we count all of the sub-agencies as part of the Department of Commerce. Whether a DOC sub-agency has an AI use case inventory, therefore, does not impact whether DOC is marked as having implemented an inventory. However, for the assessment among "large agencies with known AI uses," child-agency identified use cases were imputed to the parent agency: for example, while the ACUS Report did not identify any AI use cases by DOC at the department level, DOC was marked as having known AI use cases in the parent-level assessment because its sub-agencies had known AI use cases.
  - (ii) A parent-level measure is generally a more conservative measurement because it significantly reduces the number of small agencies assessed for compliance.

### C.2 Summary of Findings

Table 8 provides results on the filing of AI use case inventories for large, parent-level agencies that had a known use case as of 2019. The ACUS AI Report is the best available public resource for comparing the likely agencies with AI use cases. We emphasize that the difficulty of searching for and verifying agency uses of AI against the Trustworthy AI Order's requirements is precisely why disclosure is important—and, indeed, why it would be valuable even for agencies to post empty inventories so the public is made aware that the agency believes it does not have any use cases that require disclosure.

Use of the ACUS AI Report involves several nuances. First, some of the 142 agencies examined in the Report were not relevant for the use case inventory requirement given that many were either independent regulatory agencies (exempted by the terms of the Trustworthy AI Order) or no longer functional. Second, the ACUS AI Report's definition of AI deviates in small ways from the Trustworthy AI Order's definition, although the latter appears to be broader.<sup>48</sup>

<sup>&</sup>lt;sup>41</sup>One agency that the ACUS AI Report analyzed that we did not include was the Office of Medicare Hearings and Appeals, because it is not listed in the ACUS Sourcebook.
<sup>42</sup>These agencies are marked in the second column of the Full Tracker (see Appendix E.3, where the 125 agencies considered by the ACUS AI Report and relevant to the order were marked as "Yes" and agencies not considered in the ACUS AI Report were marked as "No."

<sup>&</sup>lt;sup>43</sup>The ACUS AI Report team identified 157 use cases across 64 agencies, representing around 45% of the agencies that the team canvassed.See [102, pp. 15-16]. However, some of these agencies were not included in our original 220 agencies assessed. For example, the ACUS AI Report identifies multiple AI use cases at the Securities and Exchange Commission, however the SEC is excluded from the Trustworthy AI Order as it is an independent regulatory agency under 35 U.S.C. § 3502(5). Therefore our final number of agencies with AI use cases is 49 instead of 64.

<sup>&</sup>lt;sup>44</sup>The third column of the Full Tracker, presented in Appendix E.3 marks as "Yes" only agencies for which the ACUS AI Report Team found an AI use case within the scope of the report. Agencies marked as "No" did not have a use case that the ACUS AI Report identified. Agencies marked "N/A" were excluded from this subset because they were not "large" agencies as defined by the ACUS AI Report.

<sup>&</sup>lt;sup>45</sup>The only exception was NIST's inventory, which was published separately from that of its parent agency (the Department of Commerce).

<sup>&</sup>lt;sup>46</sup>For example, the Department of the Interior's AI use case inventory discloses a U.S. Geological Survey (USGS) AI use case that was "[f]unded by the Federal Highway Administration" (FHWA), a sub-agency of the Department of Transportation, not the Department of the Interior. Despite this mention in INT's inventory, we did not mark FHWA as having an inventory because there were no FHWA use cases disclosed in DOT's inventory. The mention of FHWA in the INT inventory is an indicator of the thoroughness of USGS but cannot be assumed to indicate FHWA prepared an AI use case inventory. Similarly, that USGS disclosed in the INT inventory use cases that are a collaboration with other INT sub-agencies (namely, Fish and Wildlife Services and the Bureau of Ocean Energy Management) does not necessarily indicate that those other agencies participated in the preparation of an inventory. In contrast, because the INT inventory disclosed a non-zero number of use cases by USGS (55), we mark USGS as having an inventory.

<sup>&</sup>lt;sup>47</sup>DOJ's other two use cases were by the Justice Management Division and the Tax Division, which were not sub-agencies within our search criteria.

 $<sup>^{48}</sup>$ As noted above, the Trustworthy AI Order incorporates the FY2019 NDAA's definition of AI "as a reference point," but it anticipates that definition will be updated by

Third, the Report included anticipated uses of AI, whereas these have a more ambiguous treatment under the order: The order indicated that AI inventories should include "current and planned uses" in Section 5(b), but it also stated in Section 9(d)(iii) that it only applied to "existing and new uses of AI" and excluded "AI research and development (R&D) activities." That said, agencies that have filed AI use case inventories have commonly included use cases of AI that are under development. Fourth, the Report team searched for AI use from January to August 2019 (see [102, pp. 15-16]), and such use cases may not be operational today. If anything, however, we would expect machine learning to have been more widely adopted over the past three years.

To address these concerns, we double-checked the 23 parent agencies' identified use cases against the Trustworthy AI Order's definition and assessed whether those use cases were still plausibly in use today. When unclear, we identify additional and current use cases that would fall under the Trustworthy AI Order's inventory obligation.<sup>49</sup> In two instances, it is less clear whether agencies have active use cases.<sup>50</sup> Regardless of specific agency use cases, what this demonstrates is substantial inconsistency in how agencies have implemented the requirement.

Some of these use cases both touch on core agency functionalities and have been the subject of public disclosure. Beyond CBP's TVS

- Any artificial system that performs tasks under varying and unpredictable circumstances without significant human oversight, or that can learn from experience and improve performance when exposed to datasets.
- (2) An artificial system developed in computer software, physical hardware, or other context that solves tasks requiring human-like perception, cognition, planning, learning, communication, or physical action.
- (3) An artificial system designed to think or act like a human, including cognitive architectures and neural networks.
- (4) A set of techniques, including machine learning, that is designed to approximate a cognitive task.
- (5) An artificial system designed to act rationally, including an intelligent software agent or embodied robot that achieves goals using perception, planning, reasoning, learning, communicating, decision-making, and acting.

By "artificial intelligence," we limit our scope to the most recent forms of machine learning, which train models to learn from data. These include a range of methods (e.g., neural networks, random forests) capable of recognizing patterns in a range of types of data (e.g., numbers, text, image)—feats of recognition that, if undertaken by humans, would be generally understood to require intelligence. . . . Conceptually, AI includes a range of analytical techniques, such as rule-based or 'expert' symbolic systems, but we limit our focus to forms of machine learning. Our scope also excludes conventional forms of statistical inference (e.g., focused on causal, as opposed to predictive, inference) and forms of process automation that do not involve machine learning (e.g., an online case management system).

<sup>49</sup>For DOED, see [126]; for HUD, see [110]; and for SBA, see [140].

discussed above, we describe two further examples. First, the Internal Revenue Service's Return Review Program (RRP) uses "cuttingedge machine-learning technologies to detect, resolve, and prevent criminal and civil tax refund fraud and noncompliance" [51, 113]. While the IRS has published a privacy impact assessment stating the general purpose and data used by RRP [15], and the system has been critiqued by oversight agencies [2–4, 21], the IRS did not disclose this use case because neither it nor its parent agency published an AI use case inventory. Second, the Social Security Administration (SSA) uses an Anti-Fraud Enterprise System (AFES), an "industry-proven predictive analytics software to identify highrisk transactions for further review" [5, 42]. While SSA does not seem to have fully implemented AFES, it has published a privacy impact assessment for the initiative [6] but did not include it in its AI use case inventory [76].

Use case inventories also vary in terms of the information they provide for each listed AI use case. We highlight here examples of when inventories report performance benchmarks or other methodological details that would bear on the trustworthiness of their AI use cases. For example, one of the use cases by the U.S. Citizenship and Immigration Services (USCIS)—labeled "BET/FBI Fingerprint Success Maximization"—includes a statement estimating its efficacy but also its costs, noting a model could "catch 98% of rejected submissions" and potentially have saved "42,763 additional appointments in 2020" at cost of "forcing recapture during 11% of encounters" [43]. More attention needs to be paid to evaluation and performance assessments to enable the public, Congress, and other oversight bodies to assess the benefits and drawbacks of the use of AI.

The Department of Labor's use case with narratives about workrelated injuries and illnesses from the Survey of Occupational Injuries and Illnesses also illustrates the value of transparency regarding model development. There, employees manually classified qualitative answers to the survey into six categories, and then machine-learning algorithms were adopted to code the surveys using those labeled data as a training set. As detailed in its use case inventory, "[u]se of these autocoders subsequently expanded and coded 85% of all SOII elements for reference year (RY) 2019. This gradual increase occurred by adapting the selection criterion based on careful monitoring of the processes. This monitoring allowed the coding to expand to all six elements coded (occupation, nature, part, event, source, secondary source)" [65]. While the agency has not provided measures of time saved or accuracy, it has provided laudable details about the development process.

By contrast, the FBI's Threat Intake Processing System (TIPS), which is described as using "artificial intelligence (AI) algorithms to accurately identify, prioritize, and process actionable tips,"[61] provides less insight on evaluation. The FBI noted that it can "conduct ongoing testing on the code" and "monitor and/or audit performance," but it provides no other detail on development of performance measures.<sup>51</sup>

subsequent OMB guidance. See [25], Section 9(a). The CIO's 2021 guidance did not displace the NDAA's definition; instead, it stated that agencies "shall assess their use of AI and include criteria that aligns with the definition of AI as described in section 238(g) of the National Defense Authorization Act" [75]. That definition, in full, explains that AI means:[8]

In contrast, the ACUS AI Report provides the following discussion of its scope:([102, p. 12])

<sup>&</sup>lt;sup>50</sup>EEOC's use of AI was only obliquely mentioned in public documentation, preventing a thorough assessment of whether the AI use should be disclosed under the executive order. The original use case cited in the ACUS AI Report was derived from a recommendation about potential improvements to EEOC's "data analysis and predictive analytics activities," including "text analytics." See [102, pp. 30–31]. Other documentation suggests that EEOC's staff should be trained in the use of AI [7]. And USITC's use case posed boundary questions about whether the AI use was merely for R&D versus for future operations.

<sup>&</sup>lt;sup>51</sup>By the terms of the Trustworthy AI Order, agencies must report only "non-classified and non-sensitive use cases of AI" in their inventories, and publication should be "to the extent practicable" in light of, among other things, potential "sensitive law enforcement" information. See [25], Section 5(a), (e). Although providing information about TIPS presumably raises concerns about sensitive law enforcement decisions, we emphasize each agency's obligation to balance these concerns with the imperative of transparency, especially given that prioritization of law enforcement resources is shaped by the AI use case.

Parent-level Executive Agency	Inventory
Department of Commerce (DOC)	Yes
Department of Education (DOED)	No
Department of Health and Human Services (HHS)	Yes
Department of Homeland Security (DHS)	Yes
Department of Housing and Urban Development	No (public dis-
(HUD)	closure of no use
	cases <sup>a</sup> )
Department of Justice (DOJ)	Yes
Department of Labor (DOL)	Yes
Department of the Interior (INT)	Yes
Department of the Treasury (TRS)	No
Department of Transportation (DOT)	Yes
Department of Veterans Affairs (DVA)	Yes
Environmental Protection Agency (EPA)	Yes
Equal Employment Opportunity Commission	No
(EEOC)	
General Services Administration (GSA)	No
Legal Services Corporation (LSC)	No
National Aeronautics and Space Administration	Yes
(NASA)	
National Archives and Records Administration	No
(NARA)	
Railroad Retirement Board (RRB)	No
Small Business Administration (SBA)	No
Social Security Administration (SSA)	Yes <sup>b</sup>
United States Department of Agriculture (USDA)	Yes
United States International Trade Commission (USITC)	No
United States Postal Service (USPS)	No

Table 8: Inventory Implementation of Large, Parent-level Agencies with Known AI Use Cases

<sup>a</sup>Agencies that disclosed no use cases in their inventories were generally marked as "compliant." However, we have marked HUD as non-compliant only for the "known AI use cases" measure, given strong evidence that it has AI use cases.

<sup>b</sup>SSA only identified five AI use cases.[76]

Finally, we note that the implementation rate of the AI use case inventories is higher when focusing on the agencies enumerated in the CFO Act of 1990<sup>52</sup> or that are members of the CIO Council.<sup>53</sup> The number of CFO Act agencies that have published an inventory or a public disclosure of no relevant AI use cases is 17 (77%). The number of CIO Council member agencies that published an inventory or disclosed no use cases is 71%. Although the ACUS AI Report casts doubt on HUD's public disclosure that it has no AI use cases, we mark it as having implemented an inventory for these calculations. The relatively higher implementation rate for these agencies may illustrate that the CIO Council faces challenges in ensuring agencies not directly involved with the Council prepare and publish an AI use case inventory. Regardless, neither the Trustworthy AI Order nor the CIO's implementing guidance limited the scope of relevant agencies to those enumerated by the CFO Act or involved with the CIO Council.

### D PRE- AND POST-WHITE PAPER AI USE CASE INVENTORY TRACKER

Figure 1 details the differences in compliance with the AI use case inventory requirement before and after the publication of our white paper (see [120]) for large, parent-level agencies with known AI use cases. The right-most column is updated through July 3, 2023. The grey row includes the total number of agencies that meet the criteria for each column. The figure excludes some agencies subject to the Chief Financial Officers Act because they are independent regulatory agencies exempted from the requirement for an AI use case inventory. Compare this figure to Table 8. HUD, SBA, and USITC are marked as non-compliant because there is strong evidence that these agencies have AI use cases.

 $<sup>^{52}</sup>$ The 24 agencies listed in the CFO Act include USDA, DOC, DOED, DOE, HHS, DHS, HUD, DOJ, DOL, STAT, INT, TRS, DOT, DVA, EPA, GSA, NASA, NSF, OPM, SBA, SSA, and USAID. DOD and the NRC were excluded based on the scope of the Trustworthy AI Order, such that 22 agencies were relevant.

<sup>&</sup>lt;sup>53</sup>These agencies include USDA, DOC, DOED, DOE, HHS, DHS, HUD, DOJ, DOL, STAT, INT, TRS, DOT, DVA, EPA, GSA, NASA, NARA, NSF, OMB, OPM, SBA, SSA, and USAID.[20] Based on the scope of the Trustworthy AI Order, we excluded the Intelligence Community, NRC, and various defense-related agencies.

Large, Parent-level Agency	CFO Act Agency	Known AI Use Cases (Table 1, White Paper)	Agency Inventory (White Paper)	Agency Inventory (after publication)
	22	22	15	23
Department of Agriculture	1		Yes	Yes
Department of Commerce (DOC)	1	1	Yes	Yes
Department of Education (DOED)	1	✓	No	Yes
Department of Energy (DOE)	1		Yes	Yes
Department of Health and Human Services (HHS)	1	✓	Yes	Yes
Department of Homeland Security (DHS)	1	✓	Yes	Yes
Department of Housing and Urban Development (HUD)	1	1	Yes (but disclosed no use cases)	Yes (but disclosed no use cases)
Department of Interior (INT)	1	✓	No	Yes
Department of Justice (DOJ)	1	✓	Yes	Yes
Department of Labor (DOL)	1	✓	Yes	Yes
Department of State (STAT)	1		Yes	Yes
Department of Transportation (DOT)	1	1	No	Yes
Department of the Treasury (TRS)	1	✓	No	Yes
Department of Veterans Affairs (DVA)	1	✓	Yes	Yes
Environmental Protection Agency (EPA)	1	✓	Yes	Yes
Equal Employment Opportunity Commission (EEOC)		1	No	No
General Services Administration (GSA)	1	✓	No	Yes
Legal Services Corporation (LSC)		✓	No	No
National Archives and Records Administration (NARA)		1	No	No
National Aeronautics and Space Administration (NASA)	1	1	Yes	Yes
National Science Foundation (NSF)	1		Yes (but disclosed no use cases)	Yes (but disclosed no use cases)
Office of Personnel Management (OPM)	1		No	Yes
Railroad Retirement Board (RRB)		1	No	No
Small Business Administration (SBA)	1	1	No	Yes (but disclosed no use cases)
Social Security Administration (SSA)	1	✓	Yes	Yes
United States Agency for International Development (USAID)	1		Yes	Yes
U.S. International Trade Commission (USITC)		1	No	Yes (but disclosed no use cases)
U.S. Postal Service (USPS)		1	No	No

Figure 1: Inventory Implementation of Large, Parent-level Agencies with Known AI Use Cases Before White Paper Publication versus After (as of July 3, 2023)

### E FULL TRACKER

## E.1 Line-Level Requirements Tracker

AIES '23, August 8-10, 2023, Montréal, QC, Canada

Section	Responsible Stakeholder	Summary of Requirement	Deadline	Type of Requirement	Status of Implementation	
EO 13859 - Maintaining American Leadership in Artificial Intelligence 84 Federal Register 3967, February 11, 2019, https://www.federalregister.gov/documents/2019/02/14/2019-02544/maintaining-american-leadership-in-artificial-intelligence						
Sect. 2(a)-(e)	Implementing Agencies (see "Key Information" box)	<ul> <li>"[S]hall pursue six strategic objectives in furtherance of both promoting and protecting American advancements in AI:</li> <li>(a) Promote sustained investment in AI R&amp;D in collaboration with industry, academia, international partners and allies, and other non-Federal entities to generate technological breakthroughs in AI and related technologies and to rapidly transition those breakthroughs into capabilities that contribute to our economic and national security.</li> <li>(b) Enhance access to high quality and fully traceable Federal data, models, and computing resources, while maintaining safety, security, privacy, and confidentiality protections consistent with applicable laws and policies.</li> <li>(c) Reduce barriers to the use of AI technologies to promote their innovative application while protecting American technology, economic and national security, civil liberties, privacy, and values.</li> <li>(d) Ensure that technical standards minimize vulnerability to attacks from malicious actors and reflect Federal priorites for innovation, public trust, and public confidence in systems that use AI technologies; and develop international standards to promote and protect these priorities.</li> <li>(e) Train the next generation of American AI researchers and users through apprenticeships; skills programs; and education in science, technology, engineering, and mathematics (STEM), with an emphasis on computer science, to ensure that American workers, including Federal workers, are capable of taking full advantage of the opportunities of AI."</li> </ul>	None	Ongoing requirement	Not known because this is ongoing requirement without mandated outcomes to assess and therefore requires continued implementation	
Sect. 2(f)	Implementing Agencies	"Develop and implement an action plan, in accordance with the National Security Presidential Memorandum of February 11, 2019 (Protecting the United States Advantage in Artificial Intelligence and Related Critical Technologies) (the NSPM) to protect the advantage of the United States in Al and technology critical to United States economic and national security interests against strategic competitors and foreign adversaries."	None	Open-ended requirement	This requirement is expanded upon in section 8 of the executive order. See requirement for section 8(c) below.	

Figure 2: Line-Level Implementation Tracker, AI Leadership Order (E.O. 13859) (1 of 7)

Sect. 4(a)	Heads of Al R&D agencies (see "Key Information" box)	"[S]hall consider AI as an agency R&D priority" and "take this priority into account when developing budget proposals and planning for the use of funds in Fiscal Year 2020 and in future years." "[S]hall also consider appropriate administrative actions to increase focus on AI for 2019."	None	Ongoing requirement	Indications of sustained implementation on this ongoing requirement The National Artificial Intelligence Research and Development Strategic Plan: 2019 Update, published by the NSTC Select Committee on Al in June 2019, "highlights the key priorities for Federal investment in Al R&D" as a response to EO 13859's directive and to support the American Al initiative. The 2016 - 2019 Progress Report: Advancing Artificial Intelligence R&D, published in November 2019 by NITRD and the Select Committee, "document[s] the Nation's progress in meeting the aims of the National Artificial Intelligence Research and Development Strategic Plan: 2019 Update."
Sect. 4(b)-(b) (i)	Heads of AI R&D agencies	"[S]hall budget an amount of AI R&D that is appropriate for this prioritization" and after the submission of President's Budget request for Congress, "shall communicate plans for achieving this prioritization to the OMB Director and the OSTP Director each fiscal year through the Networking and Information Technology Research and Development (NITRD) Program."	None	Ongoing requirement	Indications of sustained implementation for this ongoing requirement (see below status for Sect. 4(b)(ii))
Sect. 4(b)(ii)	Heads of Al R&D agencies	"[S]hall identify each year, consistent with applicable law, the programs to which the AI R&D priority will apply and estimate the total amount of such funds that will be spent on each program. This information shall be communicated to the OMB Director and OSTP Director each fiscal year through the NITRD Program."	Annually within 90 days of en- actment of appropri- ations for an agency	Ongoing requirement	Indications of sustained implementation for this ongoing requirement The Networking & Information Technology, Research & Development (NITRD) Program's Supplement to the President's FY2020 Budget reported, for the first time, the federal investments in Al for FYs 2018- 2020. It also described "key programs and coordination activities" for FY2020. The 2016 - 2019 Progress Report: Advancing Artificial Intelligence R&D, although not exhaustive, also details agency Al R&D activities and how they fit within the national Al R&D priorities. Launched on August 14, 2020, and as required by the National Al Initiative Act of 2020, NITRD's Al R&D dashboard tracks nondefense Al R&D investments by agency, by program component areas, and overall. Agencies Al R&D budgets, estimates, and activities were also published in the NITRD Supplement to the President's FY2021 Budget and the NITRD and the National Al Initiative Office Supplement to the President's FY2022 Budget.
Sect. 4(c)	Heads of AI R&D agencies	"[S]hall explore opportunities for collaboration with non-Federal entities, including: the private sector; academia; non- profit organizations; State, local, tribal, and territorial governments; and foreign partners and allies, so all collaborators can benefit from each other's investment and expertise in AI R&D."	None	Ongoing requirement	Not known as this is an ongoing requirement without mandated outcomes to assess

Figure 3: Line-Level Implementation Tracker, AI Leadership Order (E.O. 13859) (2 of 7)

Sect. 5(a)	Head of all agencies	"[R]eview their Federal data and models to identify opportunities to increase access and use by the greater non-Federal AI research community in a manner that benefits that community, while protecting safety, security, privacy, and confidentiality. Specifically, agencies shall improve data and model inventory documentation to enable discovery and usability, and shall prioritize improvements to access and quality of AI data and models based on the AI research community's user feedback."	None	Open-ended requirement	Not known as no mandated public reporting on status The National AI Initiative Office has a <u>page on</u> <u>Data Resources</u> in its AI Researchers Portal that provides links to seven "quality Federal datasets that are useful for AI research." Completion of requirement likely hindered by the OMB's lack of compliance with its statutory requirements under the OPEN Government Data Act to issue guidance on making data open and establishing data inventories (see below status for Sect. 5(a)(i) on investigating barriers to access or quality limitations of Federal data).**			
Sect. 5(a)(i)	OMB Director	To "help identify datasets that will facilitate non-Federal AI R&D and testing, "the OMB Director shall publish a notice in the Federal Register inviting the public to identify additional requests for access or quality improvements for Federal data and models that would improve AI R&D and testing."	Within 90 days of EO	Time-boxed requirement	Implemented - <u>Notice of request for</u> <u>information: Identifying Priority Access or</u> <u>Quality Improvements for Federal data and</u> <u>models for Artificial Intelligence Research and</u> <u>Development (R&amp;D), and Testing</u> (July 10, 2019 - about two months after deadline)			
Sect. 5(a)(i)	OMB Director with Select Committee on Al	To "help identify datasets that will facilitate non-Federal AI R&D and testing" "shall investigate barriers to access or quality limitations of Federal data and models that impede AI R&D and testing."	Within 90 days of EO	Time-boxed requirement	Implemented - The Federal Data Strategy 2020 Action Plan, in Action 8: Improve Data and Model Resources for AI Research and Development, details the Federal Government's plan to achieve this directive, noting that that this dirctive will be completed once the OMB publishes an RFI (which was completed, see above status update). The 2020 Action Plan tracker marks this milestone as complete on August 2019, when OMB published and required comments for its RFI (see above status update). It further notes: "The RFI received 28 comments. The ability to preserve a respondent's privacy was the largest barrier identified and there were also many comments requesting more high quality and curated open data, including requests for data cleaning, normalized fields, and metadata improvements that would facilitate fitness for use and provenance tracking." However, the Federal Data Strategy 2020 Action Plan also notes that the Federal government will "address[] identified barriers by updating Federal data and source code inventory guidance for agencies to utilize in enhancing the discovery and usability of Federal data and models in AI R&D." (see below status update on Sect. 5(a)(iii)).			
**Despite	**Despite the lack of guidance for Phase II: Open Data Access and Management, some agencies (e.g., the Department of Justice, AmeriCorp) had begun							

publishing their data inventories. See <u>GAO-22-104574</u>, <u>OPEN DATA: Additional Action Required for Full Public Access</u> (December 2021). <u>OMB M-21-06</u> also refers agencies, pursuant to section 5 of the EO, to follow additional OMB guidance "regarding discovery and usability of Federal data and models for non-Federal use."

Figure 4: Line-Level Implementation Tracker, AI Leadership Order (E.O. 13859) (3 of 7)

Sect. 5(a)(ii)	OMB with its interagency councils and NSTC Select Committee on Al	"[S]hall update implementation guidance for Enterprise Data inventories and Source Code Inventories to support discovery and usability in AI R&D."	Within 120 days of EO	Time-boxed requirement	Not known as no mandated public reporting on status (expected completion ~June 2019) A document entitled Implementation Guidance to Federal Agencies Regarding Enterprise Data and Source Code Inventories (although the document cannot be identified through a search of <u>code.gov</u> ). However, an official statements about this document or its release is not readily available through an online search, nor is the document explicitly linked to the OMB. Similarly, an overview of OMB Guidance on IT authorities that is maintained by the Federal Chief Information Officiers Council (CIO Council) does not mention any guidance published in 2019 or 2020 that would fit this description (only notes a 2016 OMB Memorandum on source code guidance (OMB M-16-21)). (see https:// www.cio.gov/handbook/cio-responsibilities/ it-leadership-and-accountability/agency-it- authorities-omb-guidance/). Given these facts and the language in the document ("While feedback from the public is pending, an initial report has been completed with the input from various agencies to identify known barriers."), this Guidance does not appear to be final. Additionally, the <u>Federal Data Strategy</u> 2020 Action Plan, in Action 8: Improve Data and Model Resources for Al Research and Development, notes that OMB would "provide technical schema formats on inventories" to agencies by December 31, 2020. However the <u>2020 Action Plan tracker</u> marks this milestone as still pending OMB Open Data Plan Guidance, which the GAO's tracker associated with its December 2021 report (GAO-22-104574, OPEN DATA: Additional Action Required for Full Public Access) notes that "[i]n March 2022, OMB staff told us that action to implement this recommendation is in progress, but they have not determined a time frame for issuing the guidance." This guidance has still not been published, with the CDO Council's April 2022 Enterprise Data Inventories report noting that the guidance was forthcoming. See <u>1.14 Agency</u> . IT Authorities – OMB Guidance, also, GAO's October report GAO-21-29, Open Data
-------------------	--	---	-----------------------------	---------------------------	---

Figure 5: Line-Level Implementation Tracker, AI Leadership Order (E.O. 13859) (4 of 7)

Sect. 5(a)(iii)	Agencies	"[I]n accordance with the implementation of the Cross-Agency Priority Goal: Leveraging Federal Data as a Strategic Asset, from the March 2018 President's Management Agenda," "shall consider methods to improve the quality, usability, and appropriate access to priority data by the AI research community" and "identify associated resource implications."	Within 180 days of EO	Time-boxed requirement	Not known as no mandated public reporting on status (expected completion ~August 2019) The document entitled Implementation Guidance to Federal Agencies Regarding Enterprise Data and Source Code Inventories recommends practices for agencies to improve the quality, usability, and access to priority data for AI R&D, noting that doins so "is also consistent with the Cross Agency Priority Goal #2 of the President's Management Agenda which is leveraging data as a strategic asset." However, it is not known if this is final guidance (see above status update on Sect. 5(a)(ii)). Compliance by individual agencies is not known. The Federal Data Stratgy 2021 Action Plan, in Action 5. Public Agency Open Data Plans, also marks as a milestone that each agency will "Publish an Open Data Plan that identifies specific priority data assets, including assets that support COVID-19 response and AI R&D." These plans will be reported through agency Information Resource Management (IRM) Strategic Plans. Compliance across agencies is not known.
Sect. 5(a)(iv)	Agencies in coordination with the Senior Agency Officials for Privacy (EO 13719), heads of Federal statistical entities, Federal program managers and other relevant personnel	"In identifying data and models for consideration for increased public access,""shall identify any barriers to, or requirements associated with, increased access to and use of such data and models, including: privacy and civil liberty protections for individuals who may be affected by increased access and use. ;" "safety and security concerns;" "data documentation and formatting;" "changes necessary to ensure appropriate data and system governance;" and "any other relevant considerations."	None	Open-ended requirement	Not known because no mandated public reporting on status
Sect. 5(a)(v)	Agencies	"In accordance with the President's Management Agenda and the Cross-Agency Priority Goal: Leveraging Data as a Strategic Asset shall identify opportunities to use new technologies and best practices to increase access to and usability of open data and models, and explore appropriate controls on access to sensitive or restricted data and models, consistent with applicable laws and policies, privacy and confidentiality protections, and civil liberty protections."	None	Ongoing requirement	Not known because no mandated public reporting on status of ongoing requirement without mandated outcome
Sect. 5(b)	Secretaries of Defense, Commerce, Health and Human Services, and Energy; Administrator of NASA; Director of NSF	"[S]hall, to the extent appropriate and consistent with applicable law, prioritize the allocation of high-performance computing resources for Al-related applications through: (i) increased assignment of discretionary allocation of resources and resource reserves; or (ii) any other appropriate mechanisms."	None	Ongoing requirement	Indications of implementation of ongoing requirement (note: no mandated outcomes to assess or mandated public reporting on status) The National AI Initiative Office's <u>AI</u> <u>Researchers Portal</u> includes <u>Computer</u> <u>Resources overview</u> . There are six "Federally- supported computing infrastructure resources that are useful for AI research."

Sect. 5(c)	NSTC Select Committee on Al, in coordination with the GSA	"[S]hall submit a report to the President making recommendations on better enabling the use of cloud computing resources for federally funded AI R&D"	Within 180 days of EO	Time-boxed requirement	Implemented - The NSTC Select Committee on AI published <u>Recommendations for</u> <u>Leveraging Cloud Computing Resources</u> for Federally Funded Artificial Intelligence <u>Research and Development</u> (November 17, 2020 - about sixteen months after expected completion). In July 2022, the NSTC Select Committee publised <u>Lessons Learned from Federal Use</u> of Cloud Computing to Support Artificial Intelligence Research and Development.
Sect. 5(d)	NSTC Select Committee on Al	"[S]hall provide technical expertise to the American Technology Council on matters regarding Al and the modernization of Federal technology, data, and the delivery of digital services, as appropriate."	None	Ongoing requirement	Not known because no mandated public reporting on status of ongoing requirement without mandated outcome
Sect. 6(a)-(b)	OMB Director in coordination with OSTP Director, Director of Domestic Policy Council, and Director of NEC, and in consultation with other relevant agencies and stakeholders	"[S]hall issue a memorandum to the heads of all agencies that shall: (i) inform the development of regulatory and non- regulatory approaches by such agencies regarding technologies and industrial sectors that are either empowered or enabled by Al, and that advance American innovation while upholding civil liberties, privacy, and American values; and (ii) consider ways to reduce barriers to the use of Al technologies in order to promote their innovative application while protecting civil liberties, privacy, American values, and United States economic and national security." OMB shall issue a draft version for public comment before memorandum is finalized.	Within 180 days of EO	Time-boxed requirement	Implemented - Request for Comments on a Draft Memorandum to the Heads of Executive Departments and Agencies, "Guidance for Regulation of Artificial Intelligence Applications" was published Janaury 1, 2020 on the Federal Register for public comment - OMB M-21-06, Guidance for Regulation of Artificial Intelligence Applications was published November 17, 2020 (memorandum was supposed to be published around August 2019)
Sect. 6(c)	Heads of implementing agencies "that also have regulatory authorities"	"[S]hall review authorities and submit to OMB plans to achieve consistency with OMB memorandum described in subsection 6(a)," which became OMB M-21-06	Within 180 days of OMB memoran- dum	Time-boxed requirement	Not Implemented (expected ~May 2021) See Section III and Appendix B
Sect. 6(d)	Secretary of Commerce through the NIST Director, with participation from relevant agencies (determined by the Secretary of Commerce)	<ul> <li>"[S]hall issue a plan for Federal engagement in the development of technical standards and related tools in support of reliable, robust, and trustworthy systems that use AI technologies." Plan should be consistent with OMB Circular A-119 and include "(A) Federal priority needs for standardization; (B) identification of standards development entities in which Federal agencies should seek membership with the goal of establishing or supporting United States technical leadership roles; and (C) opportunities for and challenges to United States leadership in standardization related to AI technologies." The NSTC Select Committee on AI, private sector, academia, non-governmental entities, and other stakeholders shoudl also be consulted, as needed.</li> </ul>	Within 180 days of EO	Time-boxed requirement	Implemented - NIST published <u>A Plan for</u> <u>Federal Engagement in Developing Al</u> <u>Technical Standards and Related Tools</u> in August 2019 (on time). To develop the report, NIST also consulted the public and private sector, including through a May 2019 workshop and opportunities for public comment.

Figure 7: Line-Level Implementation Tracker, AI Leadership Order (E.O. 13859) (6 of 7)

Sect. 7(a)(i)- (ii)	Heads of implementing agencies "that also provide educational grants"	"[S]hall, to the extent consistent with applicable law, consider AI as a priority area within existing Federal fellowship and service programs," including programs focused on high school, undergraduate, and graduate fellowships, alternative education and training, that support early-career university faculty who conduct AI R&D, "scholarship for service programs," "direct commissioning programs of the United States Armed Forces," and "programs that support the development of instructional programs and curricula that encourage the integration of AI technologies into courses. " These agencies "shall annually communicate plans for achieving this prioritization to the co-chairs of the [NSTC] Select Committee [on AI]."	Annually	Ongoing requirement	Not known because no mandated public reporting on status
Sect. 7(b)	NSTC Select Committee on Al	"[S]hall provide recommendations to NSTC Committee on STEM Education regarding Al-related educational and workforce development considerations that focus on American citizens."	Within 90 days of EO	Time-boxed requirement	Not known because no mandated public reporting on statuts (expected completion ~May 2019)
Sect. 7(c)	NSTC Select Committee on Al	"[S]hall provide technical expertise to the National Council for the American Worker on matters regarding Al and the American workforce, as appropriate."	None	Ongoing requirement	Not known because no mandated public reporting on status of ongoing requirement without mandated outcome
Sect. 8(a)-(b)	Assistant to the President for National Security Affairs, in coordination with OSTP Director and recipients of the NSPM (Sect. 2(f))	"[S]hall organize the development of," and submit to the President for approval, "an action plan to protect the United States advantage in Al and Al technology critical to United States economic and national security interests against strategic competitors and adversarial nations." Note: The action plan "may be classified in full or in part, as appropriate."	Within 120 days of EO	Time-boxed requirement	Not known because no mandated public reporting on status (expected completion ~June 2019); however, there are indications that the National Security Presidential Memorandum may not itself have been issued. The Federation of American Scientists (FAS) tracks National Security Presidential Memorandums (NSPMs), Presidential Policy Directives (PDDs), and Presidential Study Directives (PSDs). It's tracking of the Trump Administration (https://irp.fas.org/offdocs/ nspm/index.html) notes that this directive was not fulfilled. Similarly, the Biden Aministration has not issues a related National Security Memoranda (NSM), to the public's knowledge (https://irp.fas.org/offdocs/nsm/index.html).
Sect. 8(c)	Agencies who are recipients of the Action Plan (Sect. 8(a))	Implement the action plan as described in subsection 8(a)-(b)	Within 120 days of EO	Time-boxed requirement	Not known because no mandated public reporting on status (expected completion ~June 2019)

Figure 8: Line-Level Implementation Tracker, AI Leadership Order (E.O. 13859) (7 of 7)

EO 13960 - Promoting the Use of Trustworthy Artificial Intelligence in the Federal Government 5 Federal Register 78939, December 8, 2020, https://www.federalregister.gov/documents/2020/12/08/2020-27065/promoting-the-use-of-trustworthy-artificial-intelligence-in-the-federal-government							
Sect. 2(b)	Agencies (defined by 44 U.S.C. 3502(1), excluding independent regulatory agencies as defined by 3502(5)); excluding the Department of Defense and agencies wholly within the Intelligence Community (see Sect. 8)	"It is the policy of the United States that responsible agencies, as defined in section 8 of this order, shall, when considering the design, development, acquisition, and use of Al in Government, be guided by the common set of Principles set forth in section 3 of this order, which are designed to foster public trust and confidence in the use of Al, protect our Nation's values, and ensure that the use of Al remains consistent with all applicable laws, including those related to privacy, civil rights, and civil liberties."	None	Ongoing requirement	Not known because no mandated public reporting on status of ongoing requirement Example agency documents that indicate implementation: - NASA's Responsible AI Plan (2022) - USAID Artificial Intelligence Action Plan (2022) - HHS Artificial Intelligence (AI) Strategy (2021) - DHS S&T Artificial Intelligence & Machine Learning Strategic Plan (2021)		
Sect. 4(a)	OMB; agencies	"To the extent" existing OMB policies that "currently address many aspect of information and information technology design, development, acquisition, and use. are consistent with the Principles set forth in this order and applicable law, these existing policies shall continue to apply to relevant aspects of AI use in Government."	None	Ongoing requirement	Not known because no mandated public reporting on status of ongoing requirement without mandated outcome		
Sect. 4(b)	OMB Director, in coordination with key stakeholders (defined by Director)	"[S]hall publicly post a roadmap for the policy guidance that OMB intends to create or revise to better support the use of AI, consistent with this order. This roadmap shall include, where appropriate, a schedule for engaging with the public and timelines for finalizing relevant policy guidance. In addressing novel aspects of the use of AI in Government, OMB shall consider updates to the breadth of its policy guidance, including OMB Circulars and Management Memoranda."	Within 180 days of EO	Time-boxed requirement	Not Implemented (expected ~June 2021) The <u>2021 Federal Data Strategy Action Plan</u> , in Action 7. Artificial Intelligence and Automation, does indicate four milestones, such as making an Algorithmic Assessment Tool publicly available for agency use and coordinating the Al use case inventories, with associated target dates. However, it does not mention policy guidance documents like Management Memorandum or OMB Circulars.		
Sect. 4(c)	Agencies; OMB	"[S]hall continue to use voluntary consensus standards developed with industry participation, where available, when such use would not be inconsistent with applicable law or otherwise impracticable. Such standards shall also be taken into consideration by OMB when revising or developing Al guidance."	None	Ongoing requirement	Not known because no mandated public reporting on status of ongoing requirement without mandated outcome		

Figure 9: Line-Level Implementation Tracker, Trustworthy AI Order (E.O. 13960) (1 of 3)

Sect. 5(a)	Federal Chief Information Officers Council (CIO Council) in coordination with other interagency bodies (as determined by CIO Council)	"[S]hall identify, provide guidance on, and make publicly available the criteria, format, and mechanisms for agency inventories of non-classified and non-sensitive use cases of Al by agencies"	Within 60 days of EO	Time-boxed requirement	Implemented - CIO Council maintains an overview of this directive and the implementation guidance on its website. In fall 2021, the CIO Council published 2021 Guidance for Creating Agency Inventories for AI Use Case, in addition to a FAQ document, Example AI Use Case Inventory Scenarios, and a template for agencies to use for creating their inventory.
Sect. 5(b)	Agencies (see Sect. 2(b) Responsible Stakeholder)	"[S]hall prepare an inventory of its non- classified and non-sensitive use cases of Al, within the scope defined by section 9 of this order, including current and planned uses, consistent with the agency's mission."	Within 180 days of CIO Council's issuance of guidance (see Sect. 5(a)); annually thereafter	Time-boxed requirement	Not known because no mandated reporting on status separate from Sect. 5(e) / see below status on Sect. 5(c)-(e)
Sect. 5(c)	Agencies (see Sect. 2(b) Responsible Stakeholder)	"As part of their respective inventories of AI use cases, agencies shall identify, review, and assess existing AI deployed and operating in support of agency missions for any inconsistencies with this order."	None	Open- ended requirement	Not known because no mandated public reporting on status separate from Sect. 5(e) / See below status on Sect. 5(c)(i) and 5(c)(ii)
Sect. 5(c)(i)	Agencies (see Sect. 2(b) Responsible Stakeholder)	"[S]hall develop plans either to achieve consistency with this order for each Al application or to retire Al applications found to be developed or used in a manner that is not consistent with this order." "These plans must be approved by the agency-designated responsible official(s), as described in section 8 of this order, within this same 120-day time period."	Within 120 days of com- pleting Al inventory	Time-boxed requirement	Not known because no mandated public reporting (expected completion ~July 2022)
Sect. 5(c)(ii)	Agencies (see Sect. 2(b) Responsible Stakeholder)	"[S]trive to implement the approved plans. subject to existing resource levels" and "in coordiantion with the Agency Data Governance Body and relevant officials from agencies"	Within 180 days of plan ap- proval	Time-boxed requirement	Not implemented, but not past deadline
Sect. 5(d)	Agencies (see Sect. 2(b) Responsible Stakeholder); CIO and CDO Councils	"[S]hall share their inventories with other agencies, to the extent practicable and consistent with applicable law and policy, including those concerning protection of privacy and of sensitive law enforcement, national security, and other protected information. This sharing shall be coordinated through the CIO and Chief Data Officer Councils"	Within 60 days of com- pleting Al inventory	Time-boxed requirement	Not known because no mandated public reporting on status (expected completion ~May/June 2022)
Sect. 5(e)	Agencies (see Sect. 2(b) Responsible Stakeholder)	"[S]hall make their inventories available to the public, to the extent practicable and in accordance with applicable law and policy, including those concerning the protection of privacy and of sensitive law enforcement, national security, and other protected information."	Within 120 days of com- pleting Al inventory	Time-boxed requirement	Not Implemented (expected ~July 2022) See Section IV and Appendix C NAIIO published 12 <u>agencies' inventories to date</u>

Figure 10: Line-Level Implementation Tracker, Trustworthy AI Order (E.O. 13960) (2 of 3)

Sect. 6	Agencies (see Sect. 2(b) Responsible Stakeholder)	"[A]re expected to participate in interagency bodies for the purpose of advancing the implementation of the Principles and the use of AI consistent with this order."	None	Ongoing requirement	Not known because no mandated outcome to assess for ongoing requirement
Sect. 6	CIO Council	"[S]hall publish a list of recommended interagency bodies and forums in which agencies may elect to participate, as appropriate and consistent with their respective authorities and missions" to fulfill the expectation that they participate in interagency bodies to advance the Al principles (see above requirement).	Within 45 days of EO	Time-boxed requirement	Not known because no mandated public reporting on status (expected completion ~February 2021)
Sect. 7(a)	Presidential Innovation Fellows (PIF) program (administed by GSA) in collaboration with other agencies	"[S]hall identify priority areas of expertise and establish an AI track to attract experts from industry and academia to undertake a period of work at an agency. These PIF experts will work within agencies to further the design, development, acquisition, and use of AI in Government, consistent with this order."	Within 90 days of EO	Time-boxed requirement	Implemented - Although the EO had not mandated public reporting on status, the <u>2022 PIF Application</u> did have a "Data Strategy and Al" track. However, this ia an ongoing requirement for PIF Fellows.
Sect. 7(b)	OPM, in coordination with GSA and other relevant agencies	"[S]hall create inventory of Federal Government rotational programs and determine how these programs can be used to expand the number of employees with Al expertise."	Within 45 days of EO	Time-boxed requirement	Not known becuase no mandated public reporting on status (expected completion ~February 2021)
Sect. 7(c)	ОРМ	"[S]hall issue a report with recommendations for how the programs in the inventory can be best used to expand the number of employees with AI expertise at the agencies. This report shall be shared with the interagency coordination bodies identified pursuant to section 6 of this order, enabling agencies to better use these programs for the use of AI, consistent with this order."	Within 180 days of creating inventory pursuant to Sect. 7(b)	Time-boxed requirement	Not known because no mandated public reporting on status (expected completion ~August/September 2021), but 2022 Federal Workforce Priorities Report, published on May 10, 2022, does not include recommendations pursuant to this EO
Sect. 8(c)	Agencies (see Sect. 2(b) Responsible Stakeholder)	"[S]hall specify the responsible official(s) at that agency who will coordinate implementation of the Principles set forth in section 3 of this order with the Agency Data Governance Body and other relevant officials and will collaborate with the interagency coordination bodies identified pursuant to section 6 of this order."	Within 30 days of EO	Time-boxed requirement	Not known because no mandated public reporting on status (expected completion ~January 2021)

Figure 11: Line-Level Implementation Tracker, Trustworthy AI Order (E.O. 13960) (3 of 3)

### Bureaucratic Challenge to U.S. AI Governance

Al in Government Act of 2020 In the Consolidated Appropriations Act, 2021 (P.L. 116-260) as Division U, Title I, December 27, 2020, <u>https://www.congress.gov/116/plaws/publ260/PLAW-116publ260.pdf</u>							
Sect. 103	GSA	Create an "AI Center of Excellence" within the GSA that shall "(1) facilitate the adoption of artificial intelligence technologies in the Federal Government; (2) improve cohesion and competency in the adoption and use of artificial intelligence within the Federal Government; and (3) carry out paragraphs (1) and (2) for the purposes of benefitting the public and enhancing the productivity and efficiency of Federal Government operations."	None	Open- ended requirement	Implemented - The Aritificial Intelligence Center of Excellence (CoE) provides a number of services (see <u>AI Services Catalog</u> ) and published an <u>"AI Guide for Government</u> " that is "A living and evolving guide to the application of artificial intelligence for the U.S. federal government."		
Sect. 103	GSA AI Center of Excellence	"The duties of the AI CoE shall include (1) regularly convening individuals from agencies, industry, Federal laboratories, nonprofit organizations, institutions of higher education, and other entities to discuss recent developments in artificial intelligence, including the dissemination of information regarding programs, pilots, and other initiatives at agencies, as well as recent trends and relevant information on the understanding, adoption, and use of artificial intelligence; (2) collecting, aggregating, and publishing on a publicly available website information regarding programs, pilots, and other initiatives led by other agencies and any other information determined appropriate by the Administrator; (3) advising the Administrator, the Director, and agencies on the acquisition and use of artificial intelligence through technical insight and expertise, as needed; (4) assist agencies in applying Federal policies regarding the management and use of data in applications of artificial intelligence; (5) consulting with agencies, including the Department of Defense, the Department of Commerce, the Department of Energy, the Department of Homeland Security, the Office of Management and Budget, the Office of the Director of National Intelligence, and the National Science Foundation, that operate programs, create standards and guidelines, or otherwise fund internal projects or coordinate between the public and private sectors relating to artificial intelligence; (6) advising the Director of Netional Intelligence by agencies; and (7) advising the Director of the Office of Science and Technology Policy on developing policy related to research and national investment in artificial intelligence."	None	Ongoing requirement	Not known because (1), (3), (4), (5), (6), and (7) are ongoing requirements without mandated outcomes to assess However, on (1), (3), (4), and (5), GSA has established a <u>Community of Practice for AI</u> that any federal employee can join to support "the practical implementation of responsible AI in the federal government" through a "monthly newsletter, events, and working groups." GSA also published in January 2022 an <u>Artificial Intelligence Governance Toolkit</u> that is intended to provide "agency leaders, privacy practitioners and others" with "a framework that addresses privacy and governance at both the organizational and system levels." On (2), GSA's Center of Excellence does publish "latest updates" but has only published three articles as of November 8, 2022 (see here)		

Figure 12: Line-Level Implementation Tracker, AI in Government Act (1 of 3)

Sect. 104(a)- (b), (d)	OMB, in coordination with OSTP Director. GSA Adminstrator, and other relevant agencies or stakeholders (determined by OMB Director)	<ul> <li>"[S]hall issue a memorandum to the head of each agency that shall <ul> <li>(1) inform the development of policies regarding Federal acquisition and use by agencies regarding technologies that are empowered or enabled by artificial intelligence, including an identification of the responsibilities of agency officials managing the use of such technology;</li> <li>(2) recommend approaches to remove barriers for use by agencies of artificial intelligence technologies in order to promote the innovative application of those technologies while protecting civil liberties, civil rights, and economic and national security;</li> <li>(3) identify best practices for identifying, assessing, and mitigating any discriminatory impact or bias on the basis of any classification protected under Federal nondiscrimination laws, or any unintended consequence of the use of artificial intelligence, including policies to identify data used to train artificial intelligence algorithms as well as the data analyzed by artificial intelligence used by the agencies; and</li> <li>(4) provide a template of the required contents of the agency plans" to comply with this memorandum.</li> <li>Sect. 104(b) requires a draft version for public comment and Sect. 104(d) requires the OMB Director issue a memorandum "every 2 years thereafter for 10 years".</li> </ul></li></ul>	Draft version for public comment not later than 180 days after enactment of Act (enacted Jan. 2021) Not later than 270 days after Act enacted	Time-boxed requirement	Not Implemented (expected draft ~July 2021 and final ~October 2021)
Sect. 104(c)	Head of agencies	"[S]hall submit to [OMB] Director and post on a publicly available page on the wesbite of the agency (1) a plan to achieve consistency with the memorandum; or (2) a written determination that the agency does not use and does not anticipate using artificial intelligence." Requires agencies submit same plan or written determination after every update the OMB Director publishes (sect. 104(d)).	No later than 180 days after OMB is- sues mem- orandum	Time-boxed requirement	Not Implemented (expected ~April 2022)

Figure 13: Line-Level Implementation Tracker, AI in Government Act (2 of 3)

Sect. 105(a)	OPM Director	<ul> <li>"[I]n accordance with chapter 51 of title 5, United States Code, the Director of the Office of Personnel Management shall—</li> <li>(1) identify key skills and competencies needed for positions related to artificial intelligence;</li> <li>(2) establish an occupational series, or update and improve an existing occupational job series, to include positions the primary duties of which relate to artificial intelligence;</li> <li>(3) to the extent appropriate, establish an estimate of the number of Federal employees in positions related to artificial intelligence, by each agency; and</li> <li>(4) using the estimate established in paragraph (3), prepare a 2-year and 5-year forecast of the number of Federal employees in positions related to artificial intelligence that each agency will need to employ."</li> </ul>	Not later than 18 months after Act enacted	Time-boxed requirement	Not Implemented (expected ~July 2022)
Sect. 105(b)	OPM Director	"[S]hall submit to the Committee on Homeland Security and Governmental Affairs of the Senate and the Committee on Oversight and Reform of the House of Representatives a comprehensive plan with a timeline to complete requirements described in subsection (a)."	Not later than 120 days after Act enacted	Time-boxed requirement	Not Implemented (expected ~May 2021)

Figure 14: Line-Level Implementation Tracker, AI in Government Act (3 of 3)

### E.2 AI Plans Tracker

Tracker of EO 13859's Requirement for Relevant Agencies to Submit to OMB Plans to Achieve Consistency with OMB M-21-06												
Agency (Abbreviation)	Reason for inclusion	Agency Plan	Guidance Webpage (www.[agencyname]. gov/guidance)	Dedicated Agency URL	Web Search	Search within Agency Website	Al.gov					
Board of Governors of the Federal Reserve System (FED)	Independent regulatory agency	No	No	No	No	No	No					
Commodity Futures Trading Commission (CFTC)	Independent regulatory agency	No	No	No	No	No	No					
Consumer Financial Protection Bureau (CFPB)	Independent regulatory agency	No	<u>Yes (redirects)</u>	No	No	No	No					
Consumer Product Safety Commission (CPSC)	Independent regulatory agency	No	No	No	No	No	No					
Department of Agriculture (USDA)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of Commerce (DOC)	Cabinet-level agency	No	No	No	No	No	No					
Department of Defense (DOD)	Cabinet-level agency	No	<u>Yes (redirects)</u>	No	No	No	No					
Department of Education (DOED)	Cabinet-level agency	No	<u>Yes (redirects)</u>	No	No	No	No					
Department of Energy (DOE)	Cabinet-level agency	Yes	Yes (redirects)	Yes	Yes	Yes	No					
Department of Health and Human Services (HHS)	Cabinet-level agency	Yes	Yes	No	Yes	Yes	No					
Department of Homeland Security (DHS)	Cabinet-level agency	<u>No</u> But see, DHS S&T AI and ML Strategic Plan	Yes	No	No	No	No					
Department of Housing and Urban Development (HUD)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of Justice (DOJ)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of Labor (DOL)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of State (STAT)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of the Interior (INT)	Cabinet-level agency	No	<u>Yes (redirects)</u>	No	No	No	No					
Department of the Treasury (TRS)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of Transportation (DOT)	Cabinet-level agency	No	Yes	No	No	No	No					
Department of Veterans Affairs (DVA)	Cabinet-level agency	Yes	Yes	Yes	Yes	No	No					
Environmental Protection Agency (EPA)	Cabinet-level agency	Yes	Yes	Yes	Yes	Yes	No					
Federal Communications Commission (FCC)	Independent regulatory agency	No	No	No	No	No	No					
Federal Deposit Insurance Corporation (FDIC)	Independent regulatory agency	No	No	No	No	No	No					
Federal Energy Regulatory Commission (FERC)	Independent regulatory agency	No	No	No	No	No	No					
Federal Housing Finance Agency (FHFA)	Independent regulatory agency	No	No	No	No	No	No					
Federal Maritime Commission (FMC)	Independent regulatory agency	No	No	No	No	No	No					
Federal Trade Commission (FTC)	Independent regulatory agency	No	No	No	No	No	No					

Figure 15: AI Plans Tracker, AI Leadership Order (E.O. 13859) (1 of 2)

Mine Enforcement Safety and Health Review Commission (FMSHRC)	Independent regulatory agency	No	No	No	No	No	No
National Labor Relations Board (NLRB)	Independent regulatory agency	No	Yes	No	No	No	No
Nuclear Regulatory Commission (NRC)	Independent regulatory agency	<u>No</u> <u>But see, Al</u> <u>Strategic Plan</u>	No	No	No	No	No
Occupational Safety and Health Review Commission (OSHRC)	Independent regulatory agency	No	No	No	No	No	No
Office of Financial Research (OFR)	Independent regulatory agency	No	No	No	No	No	No
Office of Management and Budget (OMB)	Cabinet-level agency	No	No	No	No	No	No
Office of Science and Technology Policy (OSTP)	Cabinet-level agency	No	No	No	No	No	No
Office of the Comptroller of the Currency (OCC)	Independent regulatory agency	No	No	No	No	No	No
Office of the Director of National Intelligence (ODNI)	Cabinet-level agency	No	No	No	No	No	No
Office of the United States Trade Representative (USTR)	Cabinet-level agency	No	No	No	No	No	No
Postal Regulatory Commission (PRC)	Independent regulatory agency	No	No	No	No	No	No
Securities and Exchange Commission (SEC)	Independent regulatory agency	No	No	No	No	No	No
Small Business Administration (SBA)	Cabinet	No	<u>Yes (redirects)</u>	No	No	No	No
Surface Transportation Board (STB)	Sucessor to Interstate Commerce Commission, an independent regulatory agency	No	No	No	No	No	No
United States Agency for International Development (USAID)	Inclusion in National Security Council	Yes	Yes	Yes	Yes	Yes	No

Figure 16: AI Plans Tracker, AI Leadership Order (E.O. 13859) (2 of 2)

AIES '23, August 8-10, 2023, Montréal, QC, Canada

### E.3 AI Use Case Inventories Tracker

Tracker of EO 13960s Requirement for	Al Use Case	Inventories						
Agency (Abbreviation)	"Large" Agencies	Known Al Use Case within "Large" Agencies	Inventory	Guidance Webpage	Dedicated Agency URL	Web Search	Search within Agency Website	<u>Al.gov</u>
Administrative Conference of the United States (ACUS)	No	N/A	No	Yes	No	No	No	No
Agency for Global Media (formerly Broadcasting Board of Governors)	Yes	No	No	No	No	No	No	No
Appalachian Regional Commission (ARC)	No	N/A	No	No	No	No	No	No
Barry Goldwater Scholarship and Excellence in Education Foundation (BGSEEF)	No	N/A	No	No	No	No	No	No
Chemical Safety and Hazard Investigation Board (CSHIB)	No	N/A	No	No	No	No	No	No
Corporation for National and Community Service (CNCS) (AmeriCorps)	Yes	No	No	No	No	No	No	No
Corporation for Public Broadcasting (CPB)	No	N/A	No	No	No	No	No	No
Defense Nuclear Facilities Safety Board (DNFSB)	No	N/A	No	No	No	No	No	No
Delta Regional Authority (DRA)	No	N/A	No	No	No	No	No	No
Department of Commerce (DOC)	Yes	Not for parent agency	Yes	Yes	Yes	Yes	No	Yes
Bureau of Economic Analysis (BEA)	Yes	Yes	Not listed in DOC AI Use Case Inventory	No	No	No	No	No
Bureau of Industry and Security (BIS)	No	N/A	Not listed in DOC AI Use Case Inventory	No	No	No	No	No
Economic Development Administration (EDA)	No	N/A	Not listed in DOC AI Use Case Inventory	No	No	No	No	No
International Trade Administration (ITA)	Yes	No	Yes (see DOC AI Use Case Inventory)	No	No	No	No	No
Minority Business Development Agency (MBDA)	No	N/A	Yes (see DOC AI Use Case Inventory)	No	No	No	No	No
National Institute of Standards and Technology (NIST)	Yes	No	Yes, but no identified use cases	Yes (Redirects)	Yes	Yes	No	Yes
National Ocean Service	No	N/A	Not included in DOC Al Use Case Inventory	No	No	No	No	No
National Oceanic & Atmospheric Administration (NOAA)	Yes	Yes	Yes (see DOC AI Use Case Inventory)	No	No	No	No	No
National Technical Information Service (NTIS)	No	N/A	Not included in DOC Al Use Case Inventory	No	No	No	No	No
National Telecommunications and Information Administration (NTIA)	Yes	No	Yes (see DOC Al Use Case Inventory)	No	No	No	No	No
U.S. Census Bureau (CEN)	Yes	No	Not included in DOC Al Use Case Inventory	No	No	No	No	No
U.S. Patent and Trademark Office (USPTO)	Yes	Yes	Yes (see DOC Al Use Case Inventory)	No	No	No	No	No

Figure 17: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (1 of 9)

Department of Education (DOED)	Yes	Yes	No	Yes	No	No	No	No
Federal Student Aid (OFSA)	Yes	No	No	No	No	No	No	No
Office of Elementary and Secondary Education (OESA)	No	N/A	No	No	No	No	No	No
Office of English Language Acquisition (OELA)	No	N/A	No	No	No	No	No	No
Office of Postsecondary Education (OPE)	No	N/A	No	No	No	No	No	No
Office of Special Education and Rehabilitative Services (PSER)	No	N/A	No	No	No	No	No	No
Department of Energy (DOE)	Yes	No	Yes	Yes (Redirects)	Yes	Yes	No	Yes
Loan Programs Office (LPO)	No	N/A	Not included in DOE Al Use Case Inventory	No	No	No	No	No
National Nuclear Security Administration (NNSA)	No	N/A	Not included in DOE Al Use Case Inventory	No	No	No	No	No
Office of Acquisition Management (OAM)	No	N/A	Not included in DOE Al Use Case Inventory	No	No	No	No	No
Office of Electricity (OE)	No	N/A	Yes (see DOE AI Use Case Inventory)	No	No	No	No	No
Office of Energy Efficiency and Renewable Energy (OEERE)	Yes	No	Not included in DOE Al Use Case Inventory	No	No	No	No	No
Office of Nuclear Safety Enforcement (ONSE)	No	N/A	Not included in DOE Al Use Case Inventory	No	No	No	No	No
Department of Health and Human Services (HHS)	Yes	Yes	Yes	No	No	Yes	Yes	Yes
Administration for Children and Families (ACF)	Yes	No	Not included in HHS Al Use Case Inventory	No	No	No	No	No
Administration for Community Living (ACL)	No	N/A	Not included in HHS Al Use Case Inventory	No	No	No	No	No
Administration on Aging (AOA)	No	N/A	Not included in HHS Al Use Case Inventory	No	No	No	No	No
Agency for Healthcare Research and Quality (AHRQ)	No	N/A	Yes (see HHS AI Use Case Inventory)	No	No	No	No	No
Agency for Toxic Substances and Disease Registry (ATSDR)	No	N/A	Not included in HHS Al Use Case Inventory	No	No	No	No	No
Centers for Disease Control and Prevention	Yes	Yes	Yes (see HHS AI Use Case Inventory)	No	No	Yes	No	No
Centers for Medicare and Medicaid Services (CMS)	Yes	Yes	Yes (see HHS AI Use Case Inventory)	No	No	Yes	No	No
Food and Drug Administration (FDA)	Yes	Yes	Yes (see HHS AI Use Case Inventory)	No	No	Yes	No	No
Health Resources and Services Administration (HRSA)	Yes	No	Yes (see HHS AI Use Case Inventory)	No	No	Yes	No	No
Indian Health Service (IHS)	Yes	No	Not included in HHS AI Use Case Inventory	No	No	No	No	No
Medicare Board of Trustees (Federal Supplementary Medical Insurance Trust Fund, Federal Hospital Insurance Trust Fund)	No	N/A	Not included in HHS AI Use Case Inventory	No	No	No	No	No
National Institutes of Health (NIH)	Yes	No	Yes (see HHS AI Use Case Inventory)	No	No	No	No	No
Office of Public and Indian Housing (OPIH)	Yes	No	Not included in HHS AI Use Case Inventory	No	No	No	No	No

Figure 18: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (2 of 9)

Office of the Assistant Secretary for Health (ASH)	No	N/A	Not included in HHS AI Use Case Inventory	No	No	No	No	No
Office of the National Coordinator for Health Information Technology (ONCHIT)	No	N/A	Not included in HHS Al Use Case Inventory	No	No	No	No	No
Substance Abuse and Mental Health Services Administration (SAMHSA)	Yes	Yes	Not included in HHS AI Use Case Inventory	No	No	No	No	No
Department of Homeland Security (DHS)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Domestic Nuclear Detection Office (DNDO)	No	N/A	Not listed in DHS AI Use Case Inventory	No	No	No	No	No
Federal Emergency Management Agency (FEMA)	Yes	Yes	Not listed in DHS Al Use Case Inventory	No	No	No	No	No
Federal Law Enforcement Training Center (FLETC)	Yes	No	Not listed in DHS AI Use Case Inventory	No	No	No	No	No
Transportation Security Administration (TSA)	Yes	Yes	Yes (see DHS AI Use Case Inventory)	No	No	Yes	No	No
United States Citizenship and Immigration Services (CIS)	Yes	Yes	Yes (see DHS AI Use Case Inventory)	No	No	Yes	No	No
United States Coast Guard (USCG)	Yes	Yes	Yes (see DHS AI Use Case Inventory)	No	No	Yes	No	No
United States Customs and Border Protection (CBP)	Yes	Yes	Yes (see DHS AI Use Case Inventory)	No	No	Yes	No	No
United States Immigration and Customs Enforcement (ICE)	Yes	Yes	Yes (see DHS AI Use Case Inventory)	No	No	Yes	No	No
United States Secret Service (USSS)	Yes	Yes	Not listed in DHS AI Use Case Inventory	No	No	No	No	No
Department of Housing and Urban Development (HUD)	Yes	Yes	Yes, but no use cases identified so indications of noncompliance	<u>Yes</u>	Yes	Yes	No	Yes
Federal Housing Administration (FHA)	Yes	Yes	No	No	No	No	No	No
Government National Mortgage Association (Ginnie Mae) (GNMA)	No	N/A	Νο	No	No	No	No	No
Office of Community Planning and Development (OCPD)	No	N/A	No	No	No	No	No	No
Office of Fair Housing and Equal Opportunity (OFHEO)	No	N/A	No	No	No	No	No	No
Office of Field Policy and Management (OFPM)	No	N/A	No	No	No	No	No	No
Office of Housing Counseling (OHC)	No	N/A	No	No	No	No	No	No
Office of Lead Hazard Control and Healthy Homes (OLHCHH)	No	N/A	No	No	No	No	No	No
Office of Public and Indian Housing (OPIH)	No	N/A	No	No	No	No	No	No
Department of Justice (DOJ)	Yes	Not for parent	Yes	No	No	Yes	Yes	Yes
Bureau of Alcohol, Tobacco, Firearms and Explosives (ATF)	Yes	No	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Civil Rights Division (CIVR)	Yes	No	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Community Relations Service (JCRS)	No	N/A	Not included in DOJ Al Use Case Inventory	No	No	No	No	No

Figure 19: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (3 of 9)

Drug Enforcement Administration (DEA)	Yes	No	Yes (see DOJ AI Use Case Inventory)	No	No	No	No	No
Executive Office for Immigration Review (EOIR)	Yes	Yes	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Executive Office for United States Attorneys (EOUSA)	Yes	No	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Executive Office for United States Trustees (EOUST)	Yes	No	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Federal Bureau of Investigation (FBI)	Yes	Yes	Yes (see DOJ AI Use Case Inventory)	No	No	No	No	No
Federal Bureau of Prisons (BOP)	Yes	No	Not included in DOJ AI Use Case Inventory	No	No	No	No	No
Foreign Claims Settlement Commission (FCSC)	No	N/A	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Office of Justice Programs (OJP)	Yes	Yes	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Office on Violence Against Women (OVAW)	No	N/A	Not included in DOJ AI Use Case Inventory	No	No	No	No	No
United States Marshals Service (USMS)	Yes	No	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
United States Parole Commission (USPC)	No	N/A	Not included in DOJ Al Use Case Inventory	No	No	No	No	No
Department of Labor (DOL)	Yes	Not for parent agency	Yes	No	No	Yes	Yes	Yes
Bureau of International Labor Affairs (ILAB)	No	N/A	Not included in DOL Al Use Case Inventory	No	No	No	No	No
Bureau of Labor Statistics (BLS)	Yes	Yes	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Employee Benefits Security Administration (EBSA)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Employment and Training Administration (ETA)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Mine Safety and Health Administration (MSHA)	Yes	Yes	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Occupational Safety & Health Administration (OSHA)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Office of Disability Employment Policy (ODEP)	No	N/A	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Office of Federal Contract Compliance Programs (OFCCP)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Office of Labor-Management Standards (OLMS)	No	N/A	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Office of Workers Compensation Programs (OWCP)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Pension Benefit Guaranty Corporation (PBGC)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Veterans' Employment and Training Service (VETS)	No	N/A	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Wage and Hour Division (WHD)	Yes	No	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Women's Bureau (WB)	No	N/A	Not included in DOL AI Use Case Inventory	No	No	No	No	No
Department of State (STAT)	Yes	No	Yes	No	No	Yes	Yes	Yes
Department of the Interior (INT)	Yes	Not for parent agency	Yes	No	No	Yes	Yes	No

Figure 20: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (4 of 9)

Bureau of Indian Affairs (BIA)	Yes	No	Not included in INT AI Use Case Inventory	No	No	No	No	No
Bureau of Land Management (BLM)	Yes	No	Yes (see INT AI Use Case Inventory)	No	No	No	No	No
Bureau of Ocean Energy Management (BOEM)	Yes	No	No (but two collaborations between BOEM and other INT sub-agencies noted in INT AI Use Case Inventory)	No	No	No	No	No
Bureau of Reclamation (BOR)	Yes	No	Yes (see INT AI Use Case Inventory)	No	No	No	No	No
Bureau of Safety and Environmental Enforcement (BSEE)	Yes	No	Yes (see INT AI Use Case Inventory)	No	No	No	No	No
National Indian Gaming Commission (NIGC)	No	N/A	Not included in INT AI Use Case Inventory	No	No	No	No	No
National Park Service (NPS)	Yes	No	Not included in INT AI Use Case Inventory	No	No	No	No	No
Office of Natural Resources Revenue (ONRR)	No	N/A	Not included in INT AI Use Case Inventory	No	No	No	No	No
Office of Surface Mining Reclamation and Enforcement (OSMR)	Yes	No	Not included in INT AI Use Case Inventory	No	No	No	No	No
U.S. Fish and Wildlife Service (USFWS)	Yes	Yes	No (but three collaborations between USFWS and other INT sub-agencies noted in INT AI Use Case Inventory)	No	No	No	No	No
United States Geological Survey (USGS)	Yes	Yes	Yes (see INT AI Use Case Inventory)	No	No	No	No	No
Department of the Treasury (TRS)	Yes	Not for parent agency	No	No	No	No	No	No
Alcohol and Tobacco Tax and Trade Bureau (ATTTB)	Yes	No	No	No	No	No	No	No
Bureau of Engraving and Printing (BEP)	Yes	No	No	No	No	No	No	No
Bureau of the Fiscal Service (BFS)	Yes	No	No	No	No	No	No	No
Federal Insurance Office (FIO)	Yes	No	No	No	No	No	No	No
Financial Crimes Enforcement Network (FinCEN)	No	N/A	No	No	No	No	No	No
Financial Stability Oversight Council (FSOC)	No	N/A	No	No	No	No	No	No
Internal Revenue Service (IRS)	Yes	Yes	No	No	No	No	No	No
Office of Foreign Assets Control (OFAC)	No	N/A	No	No	No	No	No	No
United States Mint (MINT)	Yes	No	No	No	No	No	No	No
Department of Transportation (DOT)	Yes	Not for parent agency	Yes	<u>Yes</u> (Redirects)	Yes	Yes	Yes	Yes
Federal Aviation Administration (FAA)	Yes	Yes	Yes (see DOT AI Use Case Inventory)	No	No	No	No	No
Federal Highway Administration (FHWA)	Yes	Yes	Not included in DOT AI Use Case Inventory (but FHWA-funded use case noted in INT inventory)	No	No	No	No	No
Federal Motor Carrier Safety Administration (FMCSA)	Yes	No	Not included in DOT Al Use Case Inventory	No	No	No	No	No
Federal Railroad Administration (FRA)	Yes	Yes	Yes (see DOT AI Use Case Inventory)	No	No	No	No	No

Figure 21: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (5 of 9)

Federal Transit Administration (FTA)	Yes	Yes	Yes (see DOT AI Use Case Inventory)	No	No	No	No	No
Maritime Administration (MA)	Yes	No	Not included in DOT AI Use Case Inventory	No	No	No	No	No
National Highway Traffic Safety Administration (NHTSA)	Yes	No	Not included in DOT AI Use Case Inventory	No	No	No	No	No
Pipeline and Hazardous Materials Safety Administration (PHMSA)	Yes	No	Not included in DOT AI Use Case Inventory	No	No	No	No	No
Saint Lawrence Seaway Development Corporation (STLSDC)	No	N/A	Not included in DOT Al Use Case Inventory	No	No	No	No	No
Department of Veterans Affairs (DVA)	Yes	Yes	Yes	No	No	Yes	Yes	Yes
Board of Veterans Appeals (BVA)	Yes	No	No	No	No	No	No	No
National Cemetery Administration (NCA)	Yes	No	No	No	No	No	No	No
Veterans Benefits Administration (VBA)	Yes	No	No	No	No	No	No	No
Veterans Health Administration (VHA)	Yes	No	No (however, VHA applications mentioned in VA AI Use Case Inventory)	No	No	No	No	No
Election Assistance Commission (EAC)	No	N/A	No	No	No	No	No	No
Environmental Protection Agency (EPA)	Yes	Yes	Yes	<u>Yes</u> (redirects)	Yes	Yes	No	Yes
Equal Employment Opportunity Commission (EEOC)	Yes	Yes	No	No	No	No	No	No
Executive Office of the President	Yes	Not for parent agency	No	No	No	No	No	No
Council of Economic Advisors (CEA)	No	N/A	No	No	No	No	No	No
Council on Environmental Quality (CEQ)	No	N/A	No	No	No	No	No	No
Office of Management and Budget (OMB)	Yes	No	No	No	No	No	No	No
Office of National Drug Control Policy (ODNCP)	No	N/A	No	No	No	No	No	No
Office of Science and Technology Policy (OSTP)	No	N/A	No	No	No	No	No	No
Office of the United States Trade Representative (USTR)	No	N/A	No	No	No	No	No	No
Export-Import Bank of the United States (EXIM)	Yes	No	No	No	No	No	No	No
Farm Credit Administration (FCA)	No	N/A	No	No	No	No	No	No
Farm Credit System Insurance Corporation (FCSIC)	Yes	No	No	No	No	No	No	No
Federal Agricultural Mortgage Corporation (FAMC) (Farmer Mac)	No	N/A	No	No	No	No	No	No
Federal Labor Relations Authority (FLRA)	No	N/A	No	No	No	No	No	No
Federal Mediation and Conciliation Service (FMCS )	No	N/A	No	No	No	No	No	No
Federal Old-Age & Survivors Insurance Trust Fund & the Federal Disability Insurance Trust Fund (FODSITF)	No	N/A	No	No	No	No	No	No

Figure 22: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (6 of 9)

						1		
Federal Retirement Thrift Investment Board (FRTIB)	No	N/A	No	No	No	No	No	No
General Services Administration	Yes	Not for parent agency	No	No	No	No	No	No
Office of Acquisition Policy	Yes	Yes	No	No	No	No	No	No
Harry S. Truman Scholarship Foundation (HSTSF)	No	N/A	No	No	No	No	No	No
Institute of Museum and Library Services (IMLS)	No	N/A	No	No	No	No	No	No
Inter-American Foundation (IAF)	No	N/A	No	No	No	No	No	No
James Madison Memorial Fellowship Foundation (JMMFF)	No	N/A	No	No	No	No	No	No
Legal Services Corporation (LSC)	Yes	Yes	No	No	No	No	No	No
Marine Mammal Commission (MMC)	No	N/A	No	No	No	No	No	No
Merit Systems Protection Board (MSPB)	No	N/A	No	No	No	No	No	No
Metropolitan Washington Airports Authority (MWAA)	No	N/A	No	No	No	No	No	No
Millennium Challenge Corporation (MCC)	No	N/A	No	No	No	No	No	No
Morris K. Udall and Stewart L. Udall Foundation (MUSUF)	No	N/A	No	No	No	No	No	No
National Aeronautics and Space Administration (NASA)	Yes	Yes	Yes	No	No	Yes	No	No
National Archives and Records Administration (NARA)	Yes	Yes	No	No	No	No	No	No
National Consumer Cooperative Bank (NCCB)	No	N/A	No	No	No	No	No	No
National Credit Union Administration (NCUA)	Yes	No	No	No	No	No	No	No
National Endowment for the Arts (NEA)	No	N/A	No	No	No	No	No	No
National Endowment for the Humanities (NEH)	No	N/A	No	No	No	No	No	No
National Institute of Building Sciences (NIBS)	No	N/A	No	No	No	No	No	No
National Mediation Board (NMB)	No	N/A	No	No	No	No	No	No
National Railroad Passenger Corporation (AMTRAK)	No	N/A	No	No	No	No	No	No
National Science Foundation (NSF)	No	N/A	Yes, but no identified use cases	Yes	Yes	No	Yes	No
National Transportation Safety Board (NTSB)	Yes	No	No	No	No	No	No	No
Office of Government Ethics (OGE)	No	N/A	No	No	No	No	No	No
Office of Personnel Management (OPM)	Yes	No	No	No	No	No	No	No
Office of Special Counsel (OSC)	No	N/A	No	No	No	No	No	No
Peace Corps (PC)	Yes	No	No	No	No	No	No	No

Figure 23: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (7 of 9)

Privacy and Civil Liberties Oversight Board (PCLOB)	No	N/A	No	No	No	No	No	No
Puerto Rico Financial Oversight and Management Board (PRFOMB)	No	N/A	No	No	No	No	No	No
Railroad Retirement Board (RRB)	Yes	Yes	No	No	No	No	No	No
Securities Investor Protection Corporation (SIPC)	No	N/A	No	No	No	No	No	No
Small Business Administration (SBA)	Yes	Yes	No	No	No	No	No	No
Social Security Administration (SSA)	Yes	Yes	Yes, but only 5 use cases identified	No	No	No	Yes	Yes
Social Security Advisory Board (SSAB)	Yes	No	No	No	No	No	No	No
State Justice Institute (SJI)	No	N/A	No	No	No	No	No	No
Tennessee Valley Authority (TVA)	Yes	No	No	No	No	No	No	No
Thrift Savings Plan (TSP)	No	N/A	No	No	No	No	No	No
U.S. Development Finance Corporation (DFC)	No	N/A	No	No	No	No	No	No
United States African Development Foundation (USADF)	No	N/A	No	No	No	No	No	No
United States Agency for International Development (USAID)	Yes	No	Yes	Yes	Yes	Yes	No	Yes
United States Department of Agriculture (USDA)	Yes	Not for parent agency	Yes	Yes	Yes	Yes	Yes	Yes
Agricultural Marketing Service (AMS)	Yes	No	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Agricultural Research Service (ARS)	Yes	Yes	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
Animal and Plant Health Inspection Service (APHIS)	Yes	No	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
Commodity Credit Corporation (CCC)	No	N/A	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Economic Research Service (ERS)	No	N/A	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
Farm Service Agency (FSA)	Yes	No	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Federal Crop Insurance Corporation (FCIC)	No	N/A	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Food and Nutrition Service (FNS)	Yes	Yes	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
Food Safety and Inspection Service (FSIS)	Yes	Yes	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Foreign Agricultural Service (FAS)	Yes	No	Not included in USDA AI Use Case Inventory	No	No	No	No	No
National Agricultural Statistics Service (NASS)	Yes	Yes	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
National Institute of Food and Agriculture (NIFA)	Yes	Yes	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
Natural Resources Conservation Service (NRCS)	Yes	Yes	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
Risk Management Agency (RMA)	Yes	No	Not included in USDA AI Use Case Inventory	No	No	No	No	No

Figure 24: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (8 of 9)

Christie Lawrence, Isaac Cui, and Daniel E. Ho

Rural Business-Cooperative Service (RBCS)	No	N/A	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Rural Housing Service (RHS)	Yes	No	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Rural Utilities Service (RUS)	No	N/A	Not included in USDA AI Use Case Inventory	No	No	No	No	No
Forest Service (USFS)	Yes	No	Yes (see USDA AI Use Case Inventory)	No	No	Yes	No	No
United States Institute of Peace (USIP)	No	N/A	No	No	No	No	No	No
United States International Trade Commission (USITC)	Yes	Yes	No	No	No	No	No	No
United States Postal Service (USPS)	Yes	Yes	No	No	No	No	No	No
United States Trade and Development Agency (USTDA)	No	N/A	No	No	No	No	No	No

Figure 25: AI Use Case Inventory Tracker, Trustworthy AI Order (E.O. 13960) (9 of 9)

Received 15 March 2023; accepted 5 May 2023

AIES '23, August 8-10, 2023, Montréal, QC, Canada